

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Axenbeck, Janna; Niebel, Thomas

Working Paper Climate protection potentials of digitalized production processes: Microeconometric evidence?

ZEW Discussion Papers, No. 21-105

Provided in Cooperation with: ZEW - Leibniz Centre for European Economic Research

Suggested Citation: Axenbeck, Janna; Niebel, Thomas (2021) : Climate protection potentials of digitalized production processes: Microeconometric evidence?, ZEW Discussion Papers, No. 21-105, ZEW - Leibniz-Zentrum für Europäische Wirtschaftsforschung, Mannheim

This Version is available at: https://hdl.handle.net/10419/250383

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



// JANNA AXENBECK AND THOMAS NIEBEL

Climate Protection Potentials of Digitalized Production Processes: Microeconometric Evidence?





Climate Protection Potentials of Digitalized Production Processes: Microeconometric Evidence?

Janna Axenbeck^{a,b,*}, Thomas Niebel^a

^a Digital Economy Department, ZEW Mannheim, P.O. Box 103443, 68034 Mannheim, Germany
 ^b Faculty of Economics, Justus-Liebig-University, Licher Str. 64, 35394 Giessen, Germany

December 16, 2021

Abstract

Although information and communication technologies (ICT) consume energy themselves, they are considered to have the potential to reduce overall energy intensity within economic sectors. While previous empirical evidence is based on aggregated data, this is the first large-scale empirical study on the relationship between ICT and energy intensity at the firm level. For this purpose, we employ administrative panel data on 28,600 manufacturing firms from German Statistical Offices collected between 2009 and 2017. Our results confirm a statistically significant and robust negative link between software capital as an indicator for the firm-level degree of digitalization and energy intensity, but the effect size is rather small. Hence, we conclude that energy intensity reductions related to the use of digital technologies are lower than expected.

Keywords: ICT, Firm-level panel data, Energy intensity improvements.

JEL Codes: D22, D24, L60, O12, O14, O33, Q40.

1. Introduction

To limit global warming, it is essential to cut down carbon emissions (IPCC, 2018). Those related to energy use, however, are still at alarmingly high levels (IEA, 2021). Especially many production processes are very energy-intensive. In 2018, the industrial sector was responsible for 37 percent of global energy use and for 24 percent of total carbon emissions (IEA, 2020b). One way to improve the environmental impact of industrial processes while remaining competitive is to decouple greenhouse gas emissions from economic growth, as stated in the European Green Deal (European Commission, 2019), for example by reducing energy intensity.

In addition to a growing need for environmental improvements, the use of digital technologies has increased strongly in recent decades. For example, new markets emerged, such as online platforms, and new channels to communicate developed, such as messaging services and digital video conferencing systems. Furthermore, "ICTs [...] heavily affected the opportunities and efficiency of how firms produce and provide goods and services" (Cardona et al., 2013, p.13). Accordingly, digital technologies – most likely – also affected energy use patterns of (manufacturing) firms and will continue to do so in the future.

However, how ICT influence environmental outcomes is ambiguous ex ante. On the one hand, digital technologies may increase firm-level energy intensity, as they consume

^{*}Corresponding author

Email address: janna.axenbeck@zew.de (Janna Axenbeck)

energy themselves. On the other hand, even though they do, they may reduce overall firm-level energy intensity due to energy efficiency improvements as well as the dematerialization of products (Berkhout and Hertin, 2004). For instance, digital technologies improve the quantity and quality of information. This allows for an improved prevention of excess production and a reduction of error rates. To put this in a nutshell, two simultaneous trends exist. Parallel to a strong need for energy intensity improvements and overall energy savings, digital technologies lead to large changes within firms. As this concurrence may contribute to or conflict with the achievement of climate targets set out in the Paris Agreement (2015), it is essential to assess how both, energy consumption and the use of digital technologies, relate to each other.

Especially grey literature assigns high climate protection potentials to digital technologies.¹ Also, previous econometric studies employing aggregated data confirm at the sectoral level that ICT can be linked to significant environmental improvements in manufacturing sectors (Schulte et al., 2016; Bernstein and Madlener, 2010). However, the use of aggregated data has several drawbacks. For instance, it does not allow for the disentanglement of dynamics within industries. Hence, it remains unobserved if actual improvements within firms exists or whether merely the composition of firms changes. Accordingly, results of studies based on aggregated data may be misleading for policy makers.

Despite the need for further research, most manufacturing countries launched programs promoting smart manufacturing such as the German "Industrie 4.0" as well as the US initiative "Smart Manufacturing Leadership Coalition" (SMLC) (Thoben et al., 2017) and emphasize its potentials for a more sustainable production.² Also the "Masterplan for a Competitive Transformation of EU Energy-intensive Industries Enabling a Climateneutral, Circular Economy by 2050" launched by the European Commission states that "digital technologies will [...] act as crosscutting enablers for industrial transformation" (European Union, 2019, p.8). Considering the number of already initiated measures, it is highly relevant from a policy perspective to correctly understand the relationship between digital technologies and energy use.

To the best of our knowledge, no large-scale microeconometric study exists yet that analyzes climate protection potentials of digitalized production processes at the firmlevel, which may provide new and more detailed insights into how the growing use of digital technologies relates to energy savings. Accordingly, this study sheds light into this relationship by employing microeconometric methods. The analysis is based on administrative panel data on 28,600 German manufacturing firms (AFiD)³ collected between 2009 and 2017 and provided by the Research Data Centres of the Statistical Offices of the Federation and the Federal States (RDC). AFiD data are of particular high quality, as reporting to the statistical offices is obligatory and the data is thoroughly checked.

Furthermore, we analyze whether firm-level software capital, as an indicator for ICT usage, affects energy intensity. The descriptive statistics already show a strong increase in software capital intensity over time, while energy intensity decreases. To ensure comparability to previous findings at the sector level, we apply a translog cost function approach for the econometric analysis. Results confirm a statistically significant link between software usage and energy intensity improvements at the firm level. However, the relationship is considerably smaller than in previous findings at the industry level.

¹For instance, see GeSI & Accenture (2015).

²The use of sensors, computing platforms, communication technology, control and simulation methods, data intensive modelling and predictive engineering within production processes is summarized as smart manufacturing (Kusiak, 2018).

³Amtliche Firmendaten für Deutschland.

We find that a 1 percent increase in software capital relates to an average decrease in energy intensity between 0.007 percent and 0.011 percent, depending on the approach how to calculate this elasticity. Results are robust to different sample restrictions as well as software capital stock modifications and the link appears to be more pronounced for energy-intensive firms and industries. Moreover, we find larger differences between than within firms. Hence, firms which have a high software capital stock on average appear to be less energy intensive on average, but when the software capital stock changes within a firm, effects have much smaller magnitude. To further analyze the robustness of our results, we conduct a reduced-form estimate with a selection of variables based on a CES production function. Respective results lead to the same conclusion, which is that the use of digital technologies is only to a small extent associated with energy intensity improvements at the firm level. As the relationship is very inelastic, we conclude that digital technologies relate to energy intensity improvements to some extent, but the effect size is not large enough to attribute substantial environmental improvements.

The remainder of this paper is structured as follows: Section 2 summarizes related literature and Section 3 presents our theoretical framework. Section 4 describes the data and provides descriptive statistics. Section 5 presents econometric specifications. Results are reported in Section 6 and discussed in Section 7. Section 8 concludes.

2. Related Literature

The European Union has declared energy efficiency improvements as a key dimension of its climate action policy (European Parliament and the Council of the European Union, 2018). "Energy efficiency is a generic term" and "refers to using less energy to produce the same amount [of output]" (Patterson, 1996, p.377). For example, a more energy efficient production process uses less energy with respect to a comparable one. A commonly-used indicator for energy efficiency is energy intensity, which is the actual amount of energy used to generate one unit of output, not necessarily considering differences in prevailing conditions, e.g., the type of product or local weather (cf. IEA, 2020*a*).

The digital transformation may influence overall energy use and intensity in various ways. Berkhout and Hertin (2004), Hilty et al. (2006) and Lange et al. (2020) develop frameworks that structure potential impact mechanisms. Based on these frameworks, the net effect on the environment consists of three different channels:

(I) Direct (Berkhout and Hertin, 2004; Lange et al., 2020) or first-order effects (Hilty et al., 2006) relate to the energy and resources consumption during the production, usage and disposal of ICT. Accordingly, direct effects have a negative environmental impact and increase energy and resource use.⁴

(II) Energy efficiency improvements (Lange et al., 2020), indirect (Berkhout and Hertin, 2004) or second-order effects (Hilty et al., 2006) refer to changes in consumption due to the application of digital technologies. Through improvements in energy efficiency as well as a substitution by dematerialized solutions, digital technologies have the potential to decrease energy intensity in different sectors.⁵ For example, Big Data allows for an improved prediction of demand and may prevent excess production, it also helps to reduce error rates. Simulation methods as well as 3D printing may drastically reduce resource and energy use associated with the design and development of new products

⁴For examples of findings on the energy consumption of YouTube see Preist et al. (2019), for the cryptocurrency Bitcoin see Stoll et al. (2019) and Corbet et al. (2021), and for data centers see Masanet et al. (2020).

 $^{{}^{5}}$ Fur instance, see Zhang et al. (2019), Ghobakhloo and Fathi (2021) or Friedrich et al. (2021) for studies that qualitatively discuss ICT-enabled energy savings in manufacturing.

(OECD, 2017; IEA, 2018). Hence, even though digital technologies consume energy, they can have a positive net effect on the environment.

(III) Structural and behavioral impacts (Berkhout and Hertin, 2004) or third-order effects (Hilty et al., 2006) describe fundamental changes associated with the use of digital technologies. For instance, a decrease in overall energy use due to energy efficiency improvements is only possible when these are not largely dampened by rebound effects (Lange et al., 2020) and digital systems are a substitute rather for than a complement to existing solutions (Berkhout and Hertin, 2004). Moreover, structural and behavioral impacts have no clear direction of impact. Therefore, Lange et al. (2020) focus on two main mechanisms with less ambiguous directions of impact: Economic growth and tertiarization. Additional consumption resulting from ICT-induced economic growth may lead to an increased energy and resource consumption. Sectoral shifts to less energy-intensive goods and services may contribute to environmental improvements.

These frameworks illustrate the complexity of the relationship between digital technologies and environmental impacts. Accordingly, determining and measuring the net impact of ICT on energy usage is not trivial. Not without reason do studies on overall trends come to different conclusions. The academic discussion presumably starts with Walker (1985), who predicts that due to productivity improvements and structural changes, the importance of electric energy will increase and overall energy efficiency will enhance. More recently, GeSI & Accenture (2015, p.8) predict based on twelve use cases that "ICT can enable a 20 percent reduction of global carbon emissions by 2030" and a large share of this reduction is attributed to the manufacturing sector. In contrast, Ferreboeuf et al. (2019) state that every year the direct energy footprint of ICT increases by 9 percent and growth could be limited to 1.5 percent, if measures to reduce the environmental impact of ICT were introduced. Also, Belkhir and Elmeligi (2018) claim that worldwide ICT-related carbon emissions could increase from approximately 3 percent in 2017 to 14 percent by 2040. Van Heddeghem et al. (2014), Andrae and Edler (2015) and Malmodin and Lundén (2018) are further studies analyzing overall trends.⁶ Most of these general studies rely on strong assumptions and not all of them are peer-reviewed. In this regard, Santarius et al. (2020) highlight the need for further research on overall trends.

To determine the net environmental impact of digital technologies it is crucial to accurately measure the size of actual energy intensity improvements. Using aggregated data to measure energy efficiency improvements within manufacturing and service industries studies come to mixed results, but tend to support the hypothesis that digital technologies are associated with a decrease in energy intensity. Using a CES production function, Collard et al. (2005) investigate the relationship between ICT and energy use in the French service sector. The authors find that electric energy intensity decreased with the diffusion of communication devices, while it increased with the use of computers and software. Applying the same approach, Bernstein and Madlener (2010) analyze the impact of ICT capital on electricity intensity in five manufacturing industries and eight EU countries from 1991 to 2005. Even though the effect seems to depend on the sector-specific production processes, the authors conclude that the diffusion of ICT is generally linked to electric efficiency improvements.

Analyzing 27 industries of ten OECD countries between 1995 and 2007 and using a translog cost function approach, Schulte et al. (2016) come to a similar conclusion. They

⁶For an overview of different (historical) studies at the sectoral or country level see Chimbo et al. (2020). For instance, applying a logistic smooth transition regression model, Lahouel et al. (2021) confirm an improved carbon efficiency related to ICT usage in Tunisia from 1970-2018 at the country level.

find that an increase in ICT capital of 1 percent is linked to a decrease in energy intensity of 0.235 percent at the sectoral level. Additionally, a sample split into manufacturing and service industries shows only significant effects for the manufacturing sector.

Unfortunately, sector-level data does not enable the analysis of dynamics within industries. It remains unknown how many energy inefficient firms drop out of the market and new, potentially ICT-intensive firms appear. This phenomenon may explain changes at the sectoral level. Furthermore, effects could only be valid for certain types of firms, e.g., larger firms that have different energy use patterns may tend to invest more in ICT. This and other issues can cause noise or misleading results as emphasized in Draca et al. (2007). Crépon and Heckel (2002) show that different methods to derive sector-level ICT capital stocks can lead to non-trivial differences in the share of ICT capital in value added. Furthermore, research that analyzes the relationship between overall capital intensity and energy use comes to different conclusions for sectoral and firm-level data (Haller and Hyland, 2014).

Comprehensive information at the firm level in manufacturing is scarce. Previous literature reviews are emphasizing the difficulty of finding objective or non-speculative studies (Ghobakhloo and Fathi, 2021; Chen et al., 2020). Exceptions are some questionnairebased surveys with non-technical self-assessments of firms. For instance, in a survey conducted in 2020, 1700 German manufacturing and services firms were asked about measures in the areas of energy efficiency and digitalization (Bertschek et al., 2020). Energy savings were the least frequently named reason for conducting ICT projects. Moreover, most manufacturing firms stated that their absolute and relative ICT-related energy use remained constant during the last three years. A survey among 65 Italian manufactures shows that firms seldom see benefits in Industry 4.0 for environmental sustainability (Brozzi et al., 2020). The largest study in this regard was conducted on behalf of the European Commission in 2021 (European Union, 2021). For this purpose, 10,006 firms were interviewed. In this survey, firms also stated that improving the environmental footprint is not the dominant motivation for implementing digital technologies. Nonetheless, 70 percent of all firms reported energy savings due to their use.

To the best of our knowledge, there is only one study at the firm-level to date that focuses on quantifiable environmental impacts. Wen et al. (2021) analyze environmental pollution measured by chemical oxygen demand (COD) and sulfur dioxide emissions (SO_2) . The authors find that an increase in ICT investments and services at the provincial-city level relates to significant reductions thereof at the firm level.

No large-scale firm-level econometric study exists yet that analyzes the impact of ICT on energy use patterns or carbon emissions. However, this might reveal new insights as "using micro data rather than industry data allows the well-documented firm level heterogeneity in productivity and investment patterns to be taken into account [...]" (Draca et al., 2007, p.113).

3. Theoretical Framework

To analyze the relationship between ICT use and energy intensity at the firm-level, we apply the same theoretical model as Schulte et al. (2016). This study is the most comprehensive at the sectoral level with results not only limited to electricity use but to general energy use and therefore best suited for a direct comparison.

The model is built on a dual translog cost function approach based on the seminal work of Christensen et al. (1973), Berndt and Wood (1975), Brown and Christensen (1980) and Berndt and Hesse (1986). We assume that the translog cost function is twice differentiable, linearly homogeneous and concave in factor prices. Different forms of capital are considered as quasi-fixed factors and materials as weakly separable. Applying

Shephard's lemma, assuming homogeneity of degree one and imposing symmetry allows estimating the following equation, in which the share of energy costs in variable costs is a function of the energy price relative to the labor price, time, output as well as ICT and non-ICT capital.⁷

$$S_E = \beta_E + \beta_{EE} ln\left(\frac{P_E}{P_L}\right) + \beta_{EK_{ICT}} ln\left(\frac{K_{ICT}}{Y}\right) + \beta_{EK_N} ln\left(\frac{K_N}{Y}\right) + \beta_{EY}^* lnY + \delta_{ET} t$$
(1)

 S_E captures the share of energy costs in variable costs (VC), which is the sum of labor and energy costs. E indicates energy, L labor and P respective prices. K_{ICT} relates to ICT capital and K_N to tangible (non-ICT) capital. Y measures total output⁸ and t the analyzed time period, which also controls for time-dependent disembodied technological progress.

The effect size of ICT on energy intensity⁹ is captured by a demand elasticity, which can be decomposed into two different effects: The first term of Equation (2) captures the effect of ICT on the share of energy costs in variable costs and the second term captures the effect of ICT on total variable costs.

$$\epsilon_{EK_{ICT}} = \frac{\partial \ln S_E}{\partial \ln K_{ICT}} + \frac{\partial \ln VC}{\partial \ln K_{ICT}} = \frac{\partial \ln E}{\partial \ln K_{ICT}}$$
(2)

Assuming that $\partial VC/\partial K_{ICT}^{10}$ equals the shadow price of ICT and rearranging Equation (2) allows measuring the demand elasticity by Equation (3) following Schulte et al. (2016). $S_{K_{ICT}}$ captures the ratio of ICT capital costs to variable costs.

$$\epsilon_{jK_{ICT}} = \frac{\beta_{jK_{ICT}}}{S_j} - S_{K_{ICT}} \tag{3}$$

In contrast, Foster-McGregor et al. (2013) and Hijzen et al. (2005) assume that the reduction in variable costs is negligible $(\partial VC/\partial K_{ICT} = 0)$. Accordingly, the demand elasticity for energy intensity following these authors is calculated as:

$$\epsilon_{jK_{ICT}} = \frac{\beta_{jK_{ICT}}}{S_j} \tag{4}$$

Both elasticities are reported in the later analysis.

4. Data

Our analysis focuses on firm-level data on the German manufacturing sector (AFiD) collected between 2009 and 2017 and provided by the Research Data Centres of the Statistical Offices of the Federation and the Federal States (RDC). Within our data, firms are assigned to the manufacturing sector if they have the highest value added in associated industries. In 2019, for instance, the manufacturing sector was responsible

⁷For a detailed description of the derivation of the model and demand elasticities see Appendix C.

 $^{{}^{8}\}beta_{EY}^{*} = \beta_{EY} + \beta_{EK_{N}} + \beta_{EK_{ICT}}$; Schulte et al. (2016) scale capital by output to be consistent with literature that measures effects of ICT on labor and output. Consequently, β_{EY} has to be modified to

 $[\]beta_{EY}^*$. It is energy intensity and not energy demand as it is controlled for output (Y).

 $^{^{10}\}partial VC/\partial K_{ICT}$ in $\partial lnVC/\partial lnK_{ICT} = (\partial VC/\partial K_{ICT})(K_{ICT}/VC).$

for 28 percent of energy demand in Germany. Approximately two-thirds were used for process heat. Mechanical energy, e.g., for operating motors or machines, represented roughly a quarter, while heating of rooms accounted only for a small share of total energy consumption (German Environment Agency, 2021). Besides, the manufacturing sector is considered as the backbone of the German economy and it is known for its efficiency.

4.1. Data Sources

We combine two AFiD datasets merged by internal identifiers from the RDC:

- (A) The AFiD-Panel Industrial Units¹¹, which contains two sub-datasets that are relevant for our analysis.
 - (A.1) The Census on Investment is used including information about investments in tangible and intangible assets. It is a full census covering all German firms in the manufacturing sector with 20 employees or more. From this survey, we retrieve our indicator for the firm-level degree of digitalization, which is software usage. Information on software investments is available since 2009. We include information on investments in property, plant and equipment from 2003 onward. This allows considering investments in tangible assets before the observation period and improves calculations of respective capital stocks. Software investments have a very high depreciation rate. Therefore, not observing such investments before the observation period is not a substantive issue, which is confirmed by a robustness check.
 - (A.2) The second applied sub-dataset is the Cost Structure Survey. It contains comprehensive annual information at the firm level about produced output as well as inputs such as energy costs, labor costs and the number of employees. The Cost Structure Survey is a stratified (partly) rotating panel. Firms with 500 employees or more are fully included in the survey, whereas firms with fewer employees are generally observed for at least four consecutive years if they are surveyed. Accordingly, our entire observation period can be divided into three sequences (2009-2011, 2012-2015, 2016-2017).
- (B) The AFiD-Module Use of Energy¹² (at the plant level) is the second applied AFiD dataset. It entails detailed information about the use of different energy sources at the plant-level. The dataset is also a full census including all German manufacturing plants with 20 employees or more. For information on firm-level energy use, we aggregate plant-level information for each firm. One minor drawback is that we do not observe the units of firms that are assigned to the service sector. Hence, when we observe software investments, it may be that they were made in a service sector facility and we cannot observe corresponding changes in energy use in that facility. However, establishments in the service sector consume a much smaller fraction of energy compared to plants in the manufacturing sector. We also do not expect to see large differences in the degree of digitalization between units within firms, as digitalization projects are most likely implemented company wide.

Additionally, we add information from several data sources. We combine AFiD with gross value added deflators from Eurostat at the two-digit industry level (NACE Rev.

¹¹DOI: 10.21242/42221.2017.00.01.1.1.0

¹²DOI: 10.21242/43531.2017.00.03.1.1.0

2 classification) to calculate real output. Annual software deflators are also taken from Eurostat. This allows us to consider real software investments and thus quality improvements in software to be taken into account. EU KLEMS data is added (also at the two-digit industry level) to receive information about capital growth rates, depreciation rates as well as tangible capital deflators. The yearly producer price index provided by the German Federal Statistical Office (Destatis) is complemented, as well as information on prices of different energy carriers. We add yearly information for national (industry) prices for the following energy sources: Electricity, natural gas, coal, heating oil, district heat, liquid gas and biomass. For a detailed overview of additionally added data, see Table A.5 in the Appendix. Sources for prices of different energy carriers are also listed here.

4.2. Variable Description

Employing the raw data described in Section 4.1, we conduct the following additional calculations to generate our model variables. The AFiD module Use of Energy entails information (in kWh) about electricity consumption as well as energetic and non-energetic use of different energy carriers, which we summarize by the following categories: Biomass, natural gas, coal, heating oil, district heat, liquid gas, and other energy sources.¹³ We define overall firm-level energy use (E) as the sum of energetic use of different energy carriers plus electricity use. Additionally, we subtract self-generated electricity by means of the listed energy carriers from electricity use to avoid double counting.



Figure 1: Mean use in GWh (left) and share of firms that use (right) different energy sources per year.

Mean use of different energy sources per year for our sample is displayed in Figure 1 (left panel). The mean fluctuates above 30 GWh and we find a small decrease in mean energy use over time. The descriptive statistics in Table 2 show that the median fluctuates around 2 GWh. Hence, the distribution of energy use is highly skewed, some firms consume far more energy than the large body of firms. To illustrate numbers, the mean firm-level energy use is more than 1900 times higher than the mean energy use of private

 $^{^{13}}$ See B.6 in the Appendix for a detailed overview of which energy carriers are included in each category.

households in 2017; the median is approximately 115 times higher.¹⁴ Figure 1 (right panel) additionally shows the share of firms that use different energy sources. All firms consume electricity. Also, more than three-quarters consume natural gas and this number is slightly increasing. Both, electricity and natural gas, also make up a large part of the mean energy use. Consequently, they can be considered as the most dominant energy sources. It should also be noted here that heating oil, which is declining, is used by more than a quarter of firms, but it accounts for only a small proportion of mean energy use. The opposite holds for coal and mineral products. They account for the third largest fraction of mean energy use, but are used by only a small amount of firms.

Energy costs $(P_E E)$ can be directly retrieved from the Cost Structure Survey. Furthermore, the analysis requires information on energy prices, which are not directly available in AFiD. Following Haller and Hyland (2014), we divide energy costs by energy use (E) to receive information on the energy price for each firm (P_E ; in \in /kWh). This approach is prone to issues resulting from misreporting. If a firm reports, for example, too low energy use, we observe too high prices. To control for outliers, we exclude the highest and lowest percentile with respect to the energy price from our analysis. The resulting price distribution is displayed in Figure F.6 in the Appendix. The energy price of most firms is between 0.02 and $0.20 \in /kWh$. Values are plausible considering industry prices for different energy sources. However, prices are endogenous as they depend, for instance, on the chosen quantity. To solve this issue, we calculate a second price variable using external energy prices (P_E [external]). We use prices of different energy sources (if available) from official statistics and weight them by the firm-level use of the respective energy source.¹⁵ Figure F.8 in the Appendix compares internal and approximated external energy prices and confirms a statistical relationship between both.¹⁶ The distribution of external prices is displayed in Figure F.7 in the Appendix, which is similar to internal energy prices, but the distribution is less skewed to the right.

Gross wages and salaries, statutory and other social costs are summarized to receive information on labor costs $(P_L L)$. The amount of full-time equivalents (L) is measured by the total number of employees adjusted for part-time employees. In the analyzed time frame, firms employ slightly more than 270 full-time equivalents on average. The yearly wage is derived by dividing labor costs by full-time equivalents. For hourly wages, we adjust values by the average yearly hours worked in 2016 in German manufacturing.¹⁷ The average hourly labor price (P_L) is $29 \in .^{18}$

Variable costs (VC) are calculated based on the sum of energy and labor costs. S_E measures the share of energy costs in variable costs, S_L the share of labor costs. The average share of energy costs in variable costs is around 0.09, which is comparable to the average sector-level share derived by Schulte et al. (2016).

Output (Y) is measured by real value added based on information specified in the Cost Structure Survey. We do not subtract energy costs to calculate value added, as we consider capital, energy and labor in our production function (KLE), but we assume materials to be weakly separable and subtract them. Output is deflated using Eurostat data on a two-digit industry level.

 $^{^{14}}$ Mean energy use of private households was 17,376 kWh in 2017, see website link (accessed 12. Nov 2021).

¹⁵See Table A.5 in the Appendix.

 $^{^{16}\}mathrm{Due}$ to a strict anonymization policy, we are not able to publish a scatter plot, as this would show individual observations.

 $^{^{17}\}mathrm{See}$ website link (accessed 13. Nov 2021).

 $^{^{18}}$ The value is a slightly higher in statistics adjusted for the overall population (website link, accessed 14. Nov 2021).

Software capital (K_{SW}) approximates the degree of firm-level digitalization and tangible capital (property, plant and equipment) represents the non-software capital stock (K_N) .¹⁹ It has to be noted here that we only account for purchased software capital and firms may also use software that is free of charge. Software and non-software capital stocks are based on investments reported in the Census on Investments. We deflate them based on Eurostat (software) and EU KLEMS data (non-software). Furthermore, the perpetual inventory method (PIM) is applied to estimate capital stocks (Griliches, 1980; Berlemann and Wesselhöft, 2014; Lutz, 2016; Dhyne et al., 2018; Löschel et al., 2019). If calculated correctly, PIM allows measuring the total productivity-relevant capital by considering next to current investments previous investments and depreciation rates. The depreciation rate of software capital in our preferred specification is 31.5 percent, as in EU KLEMS. We also calculate an average depreciation rate for non-software capital based on EU KLEMS data. Moreover, PIM requires assumptions about initial capital stocks, which are calculated based on average investments in the first three observation periods as well as depreciation and capital growth rates. Consequently, we only consider observations that are observed at least three years in a row. For a detailed description of PIM see Appendix D. Our calculated capital stocks confirm findings of Kaus et al. (2020), who analyze tangible and intangible capital within the German manufacturing sector. Software capital (as a form of intangible capital) is growing faster in comparison to tangible capital. Furthermore, both distributions of respective investments are heavily skewed and lumpy, but software investments show these characteristics to a greater extent. For instance, we find approximately 25 percent of firms without any software investments in the analyzed period. Accordingly, we add $1 \in$ to every software capital stock, as this allows for taking the logarithm when software capital stocks are zero. We will discuss this issue further in subsequent sections.

To evaluate the plausibility of estimated software capital stocks and to analyze whether they are a sufficient proxy for the firm-level degree of digitalization, we conduct the following comparisons. Firstly, we compare our results with the Survey on the Use of Information and Communication Technologies in Companies (ICT survey, 2012 - 2017), which is a stratified random sample and entails more detailed information on ICT usage.²⁰ We are able to match 16,813 observations from our sample with the ICT survey. Unfortunately, different questions are asked every year and a large share of missing values exists, so the number of observations is much lower for each survey item. Figure 2 shows mean software capital intensity, i.e., the amount of software capital used to generate one unit of output, for firms in which at least 20 percent of employees use a personal computer (PC) and for firms in which less 20 percent use a PC. Firms have a much higher software capital intensity when at least every fifth employee uses a PC. Figure 3 illustrates software capital intensity by the firm-level maximum data transmission rate. The figure depicts that the higher the Mbit/s range (the faster the internet speed), the higher also the mean software capital intensity. Consequently, we see a clear relationship between software capital and the use of other digital technologies.

Secondly, we analyze whether sectoral and regional differences with respect to software usage are plausible. Figure 4 shows average software capital intensity for different industries. Manufactures of wearing apparel (Division 14) and basic pharmaceutical products (Division 21) show the highest average software capital intensity. The pharmaceutical industry (combined with the chemical industry) was the most digital German

¹⁹Leasing capital is excluded.

 $^{^{20}{\}rm The}$ ICT survey is additionally provided by the RDC. Doi: 10.21242/52911.2012.00.00.1.1.0 - Doi: 10.21242/52911.2017.00.00.1.1.0



Figure 2: Software capital intensity by firms' PC usage.

Figure 3: Software capital intensity by maximum data transmission rate.

manufacturing industry in 2018 according to Weber et al. (2018). The high software capital intensity of the wearing apparel industry can be explained by the fact that it is a market with highly interconnected supply chains and fast changing trends. Besides, digitalization allows for an increased individualization of products, which is especially important for this industry. Furthermore, it is also intuitive that the computer industry (Division 26) uses more software than most other industries. Manufacturers of other transport equipment (Division 30) may have a comparatively high software capital intensity because related industries such as aircraft and spacecraft construction are highly innovative.



Figure 4: Average software capital intensity by industry between 2009 and 2017. The tobacco industry is excluded because of too few observations.

The geographic distribution of software capital intensity is displayed in Figure F.9 in the Appendix. The darker the blue color of the respective area, the higher the average

software capital intensity. The white area in between marks regions for which we either observe no or less than three enterprises.²¹ We find that areas with a very high software capital intensity coincide with major German cities. For example, Berlin, Munich, Dresden, Stuttgart and Hanover show very high values. As digital enterprises usually concentrate in larger cities, we consider this as a further indicator that software capital is suitable for measuring the firm-level degree of digitalization.

Additionally, the following control variables (C) are included in the analysis. We add federal state dummies as well as industry dummies on a two-digit level, dummies capturing different size classes, measured by the number of employees,²² and a dummy indicating whether a firm has multiple establishments. By means of the electric energy consumption and the ratio of electric energy costs to value added, we approximate whether firms receive a full or a partial exemption from the EEG levy. Accordingly, we include two dummies relating either to a full or partial exemption. Moreover, a dummy that controls whether a firm produces energy is included, as this may affect energy costs, as well. Last but not least, we include a dummy which is one if a firm has trading commodities.

Although AFiD data are the corner stone of many official German governmental statistics and several plausibility checks are conducted by Destatis, we find small shares of implausibly small or high values. To address this, we trim our sample by the internal labor and energy price at the 1th and 99th percentile, and winsorize all growth rates included in Equation (1) at the 0.1th and the 99.9th percentile. We also exclude firms with zero labor, energy or non-software capital use, as well as firms with a negative output. Additionally, we exploit the panel structure to identify outliers and exclude firms for which the standard deviation relative to the median of input-output ratios as well as labor and energy prices is higher than 100.

4.3. Additional Descriptive Statistics

After the described prepossessing steps our sample includes 123,362 observations, 28,600 firms in total, and on average about 13,700 firms per year (Table 1). Around 13 percent of these firms are multi-unit establishments. We point out that the last panel sequence includes slightly fewer observations than the first two. Moreover, we apply the first-difference estimator in the subsequent statistical analysis. This reduces our main estimation sample to 89,653 observations.

	Year									
Panel sequence Year	2009	$\begin{array}{c}1\\2010\end{array}$	2011	2012	2013	2 201 4	2015	2016	³ 2017	 Total
% multi-unit firms Total	$ \begin{array}{c} 13.5\% \\ 13,886 \end{array}$	$13.8\%\ 14,196$	$11.3\% \\ 13,671$	$11.3\% \\ 13,672$	$13.5\%\ 14,139$	$13.8\%\ 13,931$	$13.8\% \\ 13,581$	$\begin{smallmatrix} 13.9\% \\ 13,306 \end{smallmatrix}$	$13.9\% \\ 12,980$	$ \begin{array}{c} 13.2\% \\ 123,362 \end{array} $

Table 1: Number of observations per year.

An overview of mean, median, and standard deviation of selected variables can be found in Table 2. Values are also presented for annual change rates $(100 \times \Delta \ln)$.

 $^{^{21}\}mathrm{As}$ the RDC is not allowed to provide information at this granular level due to German data protection laws.

 $^{^{22}{\}rm Size}$ classes: 20 to 49 employees, 50 to 99 employees, 100 to 249 employees, 250 to 499 employees, 500 to 999 employees, 1000 and more.

		(1)			(2)		-
	sample	estatistics in la	avols		$100 \times \Delta l_{2}$	·*	
	mean	median	mean	median	r sd		
E	33 331 227 84	2 021 331 63	405 383 675 97	2.23	1 76	25.28	
$\frac{D}{P_{T}F}$	1848518.00	2 021 001.00	14 004 206 33	2.20	2.03	20.20	
	1 040 010.09	243 000.00	1 057 65	1.00	2.93	29.79	
	275.10	00.00	1 957.05	1.20	0.80	11.00	
$P_L L$	16 269 208.85	3778850.00	158115569.30	3.80	3.75	11.46	
P_E	0.13	0.11	0.09	1.57	1.32	35.98	
P_L	28.74	27.99	9.17	2.52	2.46	11.24	
K_{SW}	258458.48	13389.11	2703507.41	18.05	0.00	133.26	
K_N	20110799.14	3011534.63	204487549.41	2.11	-1.50	18.06	
Y	22929312.43	5060662.00	213216121.19	3.82	3.31	27.48	
Y/L	65881.09	57474.03	42847.85	2.54	1.94	29.16	
E/Y	1.07	0.38	3.86	-1.59	-1.70	36.24	
K_{SW}/Y	0.01	0.0024	0.06	14.37	-9.56	136.68	
K_N/Y	0.93	0.54	4.50	-1.64	-3.33	33.02	
VC	18117726.95	4190796.00	165630044.11	3.71	3.62	11.35	
S_L	0.91	0.94	0.10	0.05	0.02	2.24	
S_E	0.09	0.06	0.10	-0.05	-0.02	2.14	
N	123362.00		8	89 653.00			

All monetary variables in \in ; Energy is measured in kWh. *Please note that S_E and S_L are not logarithmized.

Table 2: Summary statistics of selected variables.

Column group (1) displays the sample statistics in levels. In an average manufacturing firm an employee generates approximately $65,881 \in$ output and for approximately one euro of output 1.07 kWh is used. However, half of the firms only consume 0.38 kWh or less per euro of output. Moreover, mean software capital intensity is only 0.01 and median software capital intensity is only 0.024. In comparison mean tangible capital intensity is 0.93 and median tangible capital intensity is 0.54.²³

Column group (2) shows descriptive statistics for growth rates. Mean growth rates for energy use, labor use, tangible capital and software capital are positive. Hence, the absolute use of input factors grows over time at the firm-level. Moreover, the growth rate for output is also positive. Looking at intensities, we see that output increases more strongly than most inputs. Consequently, labor, energy and capital intensity decrease over time. In contrast, software capital intensity strongly increases. It has an average growth rate of 14 percent. With 18 percent, unscaled software capital is rising even more sharply. As a clear relationship between software usage and the use other digital technologies exists, we can assume that overall ICT capital also grew strongly within the analyzed time frame. Furthermore, the descriptive statistics of growth rates point to an issue. Median software capital growth is zero and median software capital intensity growth is actually negative. Also, tangible capital (intensity) shows a negative median growth rate, which can be explained by the fact that we generally observe a highly skewed distribution of investments, a similar issue applies to software capital intensity. Moreover, zero software capital stock growth rates exist because we allow for firms with no software investments at all. Related software capital stocks remain constant at one obligatory euro. Thus, they cannot shrink and their growth rate is zero. These observations are potentially problematic for the econometric analysis. Therefore, a considerable fraction of robustness checks will address this issue. It should also be noted here that standard deviations for all logarithmized growth rates are larger than those of aggregated sectorlevel data.

Figure F.10 in the Appendix shows time trends of mean software and non-software capital as well as mean labor and energy capital divided by output relative to 2009. All

²³Ratios are comparable to aggregated EU KLEMS data.

variables decrease until 2011, which can be explained by an increase in output due to recovery after the economic crisis in 2009. Software capital intensity increases after 2011 and exceeds at the end of the observation period its level from 2009, whereas the other variables decrease in total. Moreover, the change of mean software capital should not be mistaken with the mean change of software capital.

5. Econometric Specifications

For the econometric analysis, we take first differences of Equation (1) to remove firmspecific fixed effects. Accordingly, Δu_{it} captures the time-specific firm-level deviation of firm *i*. To capture disembodied technological change at time *t*, we add a dummy variable for every year. Even though there is an unambiguous relationship between software capital intensity and the use of other digital technologies, we change ICT capital (K_{ICT}) to software capital (K_{SW}) to be accurate. Nevertheless, we assume that an increase in software capital is proportional to an overall increase in the use of digital technologies. Accordingly, percentage growth between software and other ICT capital should be comparable.

Except for the basic specification, we also add $d \in D$ control dummy variables (C). In all our specifications, we allow for clustering of observations at the firm-level when calculating the standard errors of estimates. Accordingly, Equation (1) is transformed to Equation (5).

$$\Delta S_{Eit} = \beta_{EE} \Delta ln \left(\frac{P_E}{P_L}\right)_{it} + \beta_{EK_{SW}} \Delta ln \left(\frac{K_{SW}}{Y}\right)_{it} + \beta_{EK_N} \Delta ln \left(\frac{K_N}{Y}\right)_{it} + \beta_{EY}^* \Delta lnY_{it} + \sum_{t=2010}^T \delta_{Et} t_{it} + \sum_{c=d}^D \gamma_c C_{cit} + \Delta u_{it}$$
(5)

We do not control for any further characteristics in our first specification (FD basic). However, in our second specification (FD all), which is our preferred specification, we do control for industry-specific fixed effects on a two-digit level and for firms with multiple establishments. We also add federal-state fixed effects to account for differences in wage growth as well as other differences between German federal states. Size-class dummies are included, since the cost structure may depend on the size of the firm, which is approximated by the number of employees. We additionally control for firms that may receive a full or a partial exemption from the EEG levy and for firms that produce energy by themselves. We also include a dummy indicating whether a firm has trading commodities, to control for different trends with respect to firms that outsource parts of their production due to new digital communication channels. The listed control variables will be used in all following steps.

Moreover, initial capital stocks may be unstable and investments need to be considered for a couple of periods to calculate solid capital stocks. To shed light into this issue, we will run a regression with firms observed in their third period or later in a third step (t > 3).

Additionally, we may observe a misleading correlation. If firms do not invest, their capital stock is depreciated. Hence, it decreases automatically. If especially those firms, which do not invest, increase their relative energy use, we would also measure negative capital intensity coefficients. However, this result would be deceptive. To analyze whether this is an issue with respect to software usage, we re-estimate Equation (5) and only consider observations for which the software capital stock is increasing $(\Delta ln K_{SW}(\uparrow))$.

Furthermore, endogeneity issues are a common problem in empirical studies at the firm level. We address this issue by removing time-invariant firm-specific fixed effects from the estimation. Therefore, endogeneity issues due to omitted variables are considerably reduced. For endogeneity issues caused by measurement errors of our main variable of interest, we provide various robustness checks with respect to different modifications in the calculation of software capital stocks and respective growth rates. For instance, we analyze how results change if we calculate software capital stocks assuming depreciation rates of 25, 33 and 50 percent. Also, different period lengths are employed to calculate initial capital stocks: We estimate initial software capital stocks based on two, four, and six observation periods if available.

Still, software capital stocks may be imprecisely estimated. One reason could be that a large share of firms do not report any software investments. As we impute these capital stocks in every period by an obligatory euro, they can neither rise nor shrink. However, it is not clear whether this is accurate, since firms could very well have invested in software before the observation period and their software capital stock would actually decrease in the observed time frame. Another problem may occur for firms that do not invest in periods used to calculate initial capital stocks, but start to invest afterwards. We observe then huge percentage increases in software capital stocks, since change rates from "zero" to large natural numbers are large by construction.²⁴

We conduct the following robustness checks to analyze issues with respect to "zero" software capital stocks. Firstly, we exclude all observations that have "zero" software capital stocks as well as those observations that have a software capital stock that increases from "zero" and re-estimate our model. This allows us measuring to what extent results differ when potentially problematic observations are excluded. Secondly, we look closer at observations which have "zero" software capital stocks. Hence, their software capital growth rate is zero. We do not know whether firms actually have acquired no software capital or whether they invested before the observation period and their software capital stock decreases due to depreciation. To analyze whether this makes a difference, we impute "zero" growth rates. We replace them by the logarithmic change rate that we would have observed in a firm that has software capital, but does not invest in the current period. Hence, in a further re-estimation, software capital decreases by the depreciation rate for all observations that do not invest. Thirdly, we deal with the issue that some software capital stocks increase from "zero". This may result in implausibly large growth rates. Therefore, we censor very large values that increase from "zero". We consider increases more than 5-fold as implausible, limit them at this threshold and re-estimate the model.

Moreover, we conduct a robustness check with respect to the economic crisis and exclude observations before 2011. We also estimate our model only with single-unit firms to analyze to what extent results may be biased due to inaccurately matched information in multi-unit firms. Additionally, we test whether the inclusion of tangible capital may lead to multicollinearity issues, as software investments are often complementary to them. Further, we substitute changes in software capital intensity for changes in the previous period to examine whether there are time lags in effects. In an additional specification, we include fixed effects at the sector-year level.

Furthermore, we analyze different outcomes with respect to the econometric model. Accordingly, we estimate the translog model by applying a pooled OLS and a fixed-effects estimator. Moreover, we additionally employ a hybrid Mundlak model (Mundlak, 1978; Allison, 2009). This estimator allows for the analysis of differences within and between

 $^{^{24} \}mathrm{In}$ fact they actually rise from $1 \in$ as zero values are imputed.

firms. It is a random effects estimator in which variables are decomposed into firmlevel averages (between effect) as well as their distance to the firm-level average (within effect).²⁵ Including group averages allows to relax assumptions of the random-effects estimator.

Analyses with respect to different effects regarding the sector assignment, observational period, size class and energy intensity as well as software intensity levels are displayed in Section 6.2. To further tackle endogeneity issues we aimed at conducting IV estimates, which are described in Section 6.3. In Section 6.4, we estimate a reduced form of a CES production function in order to investigate to what extent results depend on the econometric specification.

6. Estimation Results

6.1. Main Results

Table 3 shows results for the first four specifications conducted as described in the previous section. The energy intensity elasticity is calculated by both, Equation (3)[Schulte] and Equation (4) [Foster]. In addition, it is displayed for the average elasticity and the elasticity at averages of the energy cost share (S_E) and software capital cost share (S_{SW}) . Details on respective distributions are provided by Table F.7 in the $Appendix.^{26}$ The first column shows results for the baseline specification. The second column entails results for our preferred specification including a wide range of fixed effects. Both columns show similar results. All coefficients point in the same direction as in previous estimates using sector-level data. The relative energy price is positively linked to the energy cost share and the coefficient size is about the same magnitude. The coefficient for software capital intensity is negative and significant at a high threshold, but its effect size is much smaller than in previous sector-level estimates. According to the demand elasticity calculated by Equation (3), the relationship is highly inelastic. An increase in software capital of 1 percent is only associated with a 0.011 percent decrease in energy intensity. Employing Equation (4), the elasticity is only 0.007 percent. Consequently, the relationship is much smaller considering microeconometric data. In their study, Schulte et al. (2016) display Equation (3) at averages. Comparing results illustrates how large differences are between different observational levels. Applying AFiD, a 1 percent increase in software capital intensity relates to a decrease in energy intensity of 0.007 percent at firm-level averages of S_E and $S_{K_{ICT}}$, while the mentioned authors measure a decrease of 0.235 percent. One reason why we measure a smaller elasticity is the fact that we only subtract the ratio of software capital costs to variable costs and not the ratio of ICT capital cost to variable costs, which is higher by definition. However, this only explains a deviation of approximately 0.05 percentage points, thus, only a small fraction of differences. Besides, we also observe that the tangible capital and the output coefficient point into familiar directions, but have lower magnitudes than in estimates based on aggregated data. However, differences are much smaller.

The following robustness checks address issues with respect to software capital stocks. The third column displays results only including firms observed in their third period or later. Except that the output coefficient becomes insignificant, we cannot identify any notable difference to our preferred specification. The fourth column shows results only for

²⁵In the case of one independent variable, a hybrid Mundlak model would be $y_{it} = \beta_0 + \beta_W(x_{it} - \bar{x}_i) + \beta_B(\bar{x}_i) + \varepsilon_{it}$

 $[\]bar{x}_i) + \beta_B(\bar{x}_i) + \varepsilon_{it}$ ²⁶The capital compensation or shadow price to approximate $S_{K_{SW}}$ is derived by the user costs of capital calculated with EU KLEMS data and it is assumed to be $0.4 \in$.

	(1)	(2)	(3)	(4)
	FD Basic	FD ALL	t > 3	$\Delta ln K_{SW}(\uparrow)$
	ΔS_E	ΔS_E	ΔS_E	ΔS_E
$\Delta \ln(\frac{P_E}{P_L})$	0.0285***	0.0284***	0.0295***	0.0251***
	(62.02)	(169.56)	(51.02)	(35.08)
$\Delta \ln(\frac{K_{SW}}{Y})$	-0.000245***	-0.000238***	-0.000206***	-0.000214^{***}
	(-5.20)	(-5.19)	(-4.47)	(-4.11)
$\Delta \ln(\frac{K_N}{Y})$	-0.0013***	-0.0015***	-0.0013**	-0.0011
	(-3.43)	(-4.42)	(-3.27)	(-1.31)
$ riangle \ln(Y)$	0.0017^{**}	0.0013***	0.0010	0.0014
	(3.21)	(3.32)	(1.48)	(1.43)
Year	х	х	х	x
Economic sector		x	х	x
Multi-unit		x	х	x
Federal state		x	х	x
Size class		x	х	x
EEG exemption		x	х	x
Producer		х	х	x
$\overline{\epsilon}_{EK_{SW}}$ (Schulte)	-0.0111	-0.0110	-0.0102	-0.0137
$\epsilon_{EK_{SW}}$ at $\overline{S}_{E,K_{SW}}$ (Schulte)	-0.0068	-0.0068	-0.0066	-0.0098
$\overline{\epsilon}_{EK_{SW}}$ (Foster)	-0.0069	-0.0067	-0.0059	-0.0062
$\epsilon_{EK_{SW}}$ at $\overline{S}_{E,K_{SW}}$ (Foster)	-0.0026	-0.0026	-0.0022	-0.0024
Observations	89653	89653	59405	25715
Adjusted \mathbb{R}^2	0.267	0.271	0.290	0.250

Table 3: First-difference estimation results of Equation (5).

t statistics in parentheses. First-difference estimation. Clustered standard errors.

* p < 0.05, ** p < 0.01, *** p < 0.001. $\bar{\epsilon}_{EK_{SW}}$ displays the average demand elasticity for energy intensity for equation (3) and for equation (4). $\epsilon_{EK_{SW}}$ at $\overline{S}_{E,K_{SW}}$ displays the demand elasticity for energy intensity at averages of S_E and $S_{K_{SW}}$.

increasing software capital stocks. The software capital coefficient is again significant at a high threshold and the effect size is comparable to our preferred specification. However, non-software capital and output are insignificant. Both Equation (3) elasticities are slightly larger. This deviation is due to a higher average ratio of software capital costs to variable costs for the subsample used in this estimation. Also, note that slight differences in Columns (3) and (4) might stem from dramatically reduced sample sizes as well as from a selective consideration of observations.

Estimates employing different depreciation rates for the software capital stock can be found in Table G.8 in the Appendix. Results show that changes in the depreciation rate of the software capital stock only lead to marginal differences between coefficients. Hence, results appear to be robust in this regard. Results for different maximum lengths of observation periods considered for the initial capital stock calculation are displayed in Table G.9 in the Appendix. We find slight differences for initial software capital stocks that include two as well as up to six periods. For initial stocks based on two periods, we find effects that are marginally smaller. For initial stocks based on up to six periods, the effect size is slightly larger and the software capital coefficient becomes -0.0003. However, we do not consider this deviation to be large enough to have an effect on the economic interpretation of results, which would be that a 1 percent increase in software capital intensity relates only to marginal energy intensity improvements within firms.

Result with respect to issues related to "zero" software capital stocks can be found in Table G.10 in the Appendix. In the first column, we exclude observations that have "zero" software capital stocks as well as observations that have a software capital stock that increases from "zero". The effect size of the software capital coefficient is comparable to our preferred specification, however, it is only significant at the 10%-level. Consequently, even though the exclusion of potentially problematic observations does only marginally alter the coefficient size, we have to acknowledge that the relationship is now significant at a lower threshold. The second column displays results for imputed depreciation rates for "zero" software capital stocks that would occur if a firm had invested in previous periods. We find that this modification does not notably affect results and the coefficient of interest is comparable to baseline results. Hence, is does not make a difference whether we depreciate "zero" software capital stocks or not. The last two columns relate to estimates in which increases from "zero" software capital stocks are limited to a threshold. In Column (3), "zero" software capital stocks are included and in Column (4) excluded. Results of both columns are comparable and the coefficient slightly increases, but not substantially. To sum up, different treatments of growth rates related to "zero" software capital stocks only marginally affect results, i.e. they can hardly be the reason why we find smaller effects in comparison to aggregated estimates.

Table G.11 in the Appendix shows effects for single-unit firms (Column 1) and estimation results, in which only observations after 2011 are considered (Column 2). The restricted estimates are consistent with our baseline results. Both software coefficients point into a negative direction and are significant, but software coefficients are slightly smaller for both restricted samples. Moreover, our results are also robust with respect to the exclusion of tangible capital (Column 3). We additionally estimate the influence of lagged software capital and do not find any effect of an increase in the software capital stock in the previous period on current changes in the share of energy costs in variables costs (Column 4). Including sector-year level fixed effects does not affect baseline results notably (Column 5).²⁷

Table G.12 shows results with respect to different econometric models. Column (1) displays results for the pooled OLS estimator. The software capital coefficient becomes nearly seven times larger, but also other coefficients change. Column (2) provides results for the fixed-effects estimator. The coefficients of the fixed effects model are comparable to coefficients in Table 3. The coefficients point in the same direction and have more or less the same magnitude – except tangible capital, which is insignificant. The consistency of both models supports the validity of both approaches. Differences between the pooled OLS estimator and the first-difference or the fixed-effects estimator indicate a large omitted variable bias if individual-specific effects remain unconsidered. Column (3) shows results for the hybrid Mundlak model, in which the overall effect is decomposed into a between and within effect. The significance of the coefficient for average software capital confirms that individual-specific effects are correlated with the dependent variable in this regard. Results also illustrate that between effects – differences between firms – are much larger than within effects – changes within a firm. In other words, firms which

 $^{^{27}}$ Besides, we also performed an estimation in which we replaced software capital intensity with software investment and measured a comparable relationship. The results are available from the authors upon request.

have on average a higher software capital intensity tend to have lower relative energy costs on average. However, a 1 percent increase in software capital intensity within a firm only relates to a marginal reduction of relative energy costs or energy intensity.

To summarize results, an increase in software capital intensity is associated with a decrease in relative energy demand, but the relationship has a much smaller magnitude than previous industry-level estimates suggests. Effect sizes at the firm-level are robust with respect to various econometric specifications.

6.2. Robustness with Respect to Firm-level Characteristics

In this section, we shed light into the robustness of estimates with respect to different manufacturing industries, energy and software-capital-intensity levels, CSS waves as well as size classes.

6.2.1. Industries

To analyze to what extent results differ for certain types of firms, we first split our sample based on sector affiliations. Accordingly, we individually fit Equation (5) for different industries at the two-digit level (NACE classification). Estimation coefficients for software capital intensity by industry are displayed in Figure 5. The colored dots mark respective estimation coefficients and the corresponding lines represent confidence intervals at the 95%-level. If estimated independently, the software coefficient is negative but insignificant for most industries. However, it has significant negative effects for manufacturers of paper and paper products (Division 17), chemicals and chemical products (Division 20), other non-metallic products – including the cement industry – (Division 23), basic metals – including iron and steel industry – (Division 24), electrical equipment (Division 27) as well as for the repair and installation industry (Division 33). Most of these sectors show a very high energy intensity.²⁸ Consequently, a reduction in the share of energy costs in variable costs may be driven by sectors that are highly energy intensive, such as the non-metallic products and the basic metals industry. Hence, although their magnitude is small, effects at least appear to be more pronounced in industries with potentials to save larger amounts of energy.

6.2.2. Energy and Software Capital Intensity

To shed more light into the extent to which results diverge for different energy and software capital intensity levels, we split our sample into quartiles with respect to firmlevel averages of both variables respectively. Moreover, we estimate the translog model by employing a fixed-effects estimator. We do this because the fixed-effects estimator considers divergences from means. This allows for the analysis of effect differences regarding the distance to different average intensity levels.²⁹ Results are displayed in Figure H.11. The blue bars correspond to quartiles of firms with the lowest average energy or software capital intensity.

The left panel relates to different levels of average energy intensity. For the two quartiles with the lowest average energy intensity levels, the point estimates of the software capital coefficient ($\beta_{EK_{SW}}$) do barely differ from zero. For the two quartiles with the highest energy intensity, however, the magnitude of the coefficient increases and the relationship is significantly negative. Hence, the higher the average energy intensity level, the more pronounced reductions in the share of energy costs in variable costs.

 $^{^{28}}$ See website link (accessed 13. Oct 2021).

 $^{^{29}\}mbox{Please}$ note here that this is particularly relevant for different levels of average software capital intensity.



Figure 5: Heterogeneous effects - Sector level estimations (two-digit NACE level). Tobacco industry excluded because of too few observations.

The right panel relates to different levels of average software capital intensity. The quartile with the lowest average software capital intensity has a very large confidence interval. This can be explained by the fact that we observe a large share of "zero" software capital stocks, which are included here. Moreover, we do not observe a clear trend with respect to higher levels of software capital intensity. Effects of the third and the fourth quarter appear to be more pronounced, but differences are not substantial.

6.2.3. Time

Moreover, the relationship between software usage and energy intensity improvements may generally change over time. For instance, new energy-efficient technologies may appear or energy-saving potentials of already existing technologies may become exhausted. Also, effects may differ by CSS samples, as it is a rotating panel, in which a considerable share of firms is only observed for four consecutive years. Hence, the composition of firms partly changes every four years. We investigate differences with respect to the observed time frame by always estimating our model for a time frame of four years, but rotating forward by one year for each estimation. Results are displayed in Figure H.12. Software capital coefficients are always negative and significantly different from zero. Moreover, the magnitude slightly declines between the first two samples including the earliest observations and then continuously rises. However, the increase is very small. As a consequence, the difference between early and later observed samples is only marginal and we cannot find any substantial differences with respect to time.

6.2.4. Size Classes

We also estimate the translog model individually for different size classes with respect to the number of employees.³⁰ Results are presented in Figure H.13. We cannot find a linear pattern for different size classes. The software capital coefficient is relatively small and insignificant for small firms between 20 and 49 employees as well as for larger midsized firms between 100 and 249 employees. Effects are more pronounced and significant for smaller mid-sized firms between 50 and 99 employees as well as for large firms with 250 employees or more.

6.3. IV estimates

Even though we account for a large share of endogeneity issues by removing timeinvariant firm-specific effects from the estimation, we still may observe biased coefficients. For instance, software capital intensity could be endogenous due to a simultaneous relationship or a still not detected measurement error. To analyze this issue, we include information on household broadband availability (HBA) at the municipality level and use different levels of maximum available data transmission rates as instruments for software usage. A similar approach has been conducted for example by Bertschek et al. (2013). Unfortunately, HBA instruments are weak and yield unreliable estimates. Therefore, we abstained from including results in the analysis.³¹

Moreover, endogenous control variables do not lead to biased coefficients when uncorrelated with the variable of interest, but they do if they relate to each other (Frisch and Waugh, 1933). In particular, the relationship between output and software capital could potentially bias results, because both variables may highly depend on each other. A

 $^{^{30}{\}rm We}$ use four size classes instead of six to ensure that the number of observations in every subsample is comparable.

³¹Results are available from the authors upon request.

similar problem may exist with respect to the energy-labor-price ratio, as the use of software usually requires skills that are in high demand. To test whether these issues affect results, we conduct the following IV estimates as further robustness checks. Respective results are displayed in Table G.13.

For the analysis of endogeneity problems related to our output indicator, a problem arises because the variable is included in the model and also both capital stocks are scaled by output. However, from an econometric perspective it is not necessary to scale capital stocks by output since we already control for it.³² Accordingly, we rearrange Equation (1) and do not scale both capital stocks by output anymore. Hence, we now estimate β_{EY} instead of β_{EY}^* (see Section 3 or Appendix C). The translog model is re-estimated and displayed in Column (1). It is straightforward to see that this modification barely affects software and tangible capital coefficients. In a second step, we instrument output by a firm's market share in terms of sales. Market shares are calculated using four-digit and two-digit industry levels and employing the Census on Investments, accordingly two different instruments are used. Additionally, we calculate market shares only if at least three observations per industry are available. Hence, we exclude a small share of observations from the estimation. Results are displayed in Column (2). The output coefficient gently increases, but the software capital coefficient is barely affected. In a last step, we instrument the energy-labor-price ratio by an exogenous energy price variable. To calculate the exogenous energy price, we use prices of different energy sources (if available) from official statistics and weight them by the individual use of the respective energy source.³³ Results are displayed in Column (3), in Column (4) output is instrumented as well. The effect size of the price coefficient decreases and it is now significant at lower threshold, but the software capital coefficient is not affected and comparable to the baseline specification. Besides, test statistics of tests for underidentification and weak identification as well as the Sargan–Hansen test indicate that the exogenous energy price and market shares are appropriate instruments.

6.4. Estimate of a reduced CES production function

To test the robustness of results derived by the translog model, we estimate a second model in which we use energy intensity directly as a dependent variable. For this purpose, we apply a nested CES production function approach with 3-inputs (K,L; E). This approach has also been used previously in studies at the sectoral level (Bernstein and Madlener, 2010; Collard et al., 2005). Here, energy intensity is modelled as a function of the energy-related level of technology (A), the energy price relative to a general input price level (P_E/P_{PPI}), an elasticity (σ) as well as a constant (ω) (Collard et al., 2005; Van der Werf, 2008; Lagomarsino, 2020; Bernstein and Madlener, 2010).

$$ln\left(\frac{E}{Y}\right)_{it} = \sigma ln(\omega) - \sigma ln\left(\frac{P_E}{P_{PPI}}\right)_{it} + (1 - \sigma)lnA_{it} \tag{6}$$

Following Collard et al. (2005), we assume that the energy-related level of technology evolves as:

$$lnA_{it} = \theta_0 + \theta_{ICT} ln \left(\frac{K_{ICT}}{K_N}\right)_{it} + \theta_t t_{it}$$
⁽⁷⁾

 $^{^{32}}$ We scale software capital by output in our preferred specification to be consistent with Schulte et al. (2016).

 $^{^{33}}$ See Table A.5 in the Appendix for more details on data sources. The distribution of the exogenous price variable and its relationship to the potentially endogenous energy price variable is displayed in Appendix Appendix F.1.

To analyze whether effects differ between production function approaches, we plug Equation (7) in Equation (6), take first differences and estimate a reduced form as illustrated in Equation (8). The general input price level is measured by the producer price index³⁴, which is retrieved at a two-digit industry level from Destatis. Technological progress t is measured by time dummies.

$$\Delta ln \left(\frac{E}{Y}\right)_{it} = \Delta \beta_{\frac{P_E}{P_P P I}} ln \left(\frac{P_E}{P_P P I}\right)_{it} + \Delta \beta_{\frac{K_{ICT}}{K_N}} ln \left(\frac{K_{ICT}}{K_N}\right)_{it} + \sum_{t=2010}^{T} \delta_{Et} t_{it} + \sum_{c=d}^{D} \gamma_c C_{it} + \Delta u_{it}$$

$$(8)$$

Results are presented in Table 4. Three specifications are estimated. In the first column, the specification is equivalent to our preferred specification in Table 3.

(1)	(2)	(3)
FD ALL	Excluding problematic observations	$K_{SW}(\uparrow)$
$\Delta \ln E/Y$	$\Delta \ln E/Y$	$\Delta \ln E/Y$
-0.446***	-0.462***	-0.481***
(-58.43)	(-50.21)	(-34.27)
-0.00289***	-0.00456*	-0.00239**
(-3.71)	(-2.31)	(-2.68)
x	х	x
x	х	х
х	х	х
x	х	х
x	х	х
x	х	х
x	х	x
89267	64991	25609
0.228	0.231	0.252
	(1) FD ALL $\Delta ln E/Y$ -0.446*** (-58.43) -0.00289*** (-3.71) x x x x x x x x x x x x x	$\begin{array}{c ccc} (1) & (2) \\ FD ALL & Excluding \\ problematic \\ observations \\ \hline \Delta \ln E/Y & \Delta \ln E/Y \\ \hline -0.446^{***} & -0.462^{***} \\ (-58.43) & (-50.21) \\ -0.00289^{***} & -0.00456^{*} \\ (-3.71) & (-2.31) \\ \hline x & x \\ (-3.71) & (-2.31) \\ \hline x & x \\ x & x$

Table 4: First-difference results of Equation (8).

t statistics in parentheses; First-difference estimation; Clustered standard errors; * p < 0.05, ** p < 0.01, *** p < 0.001.

The adjusted R^2 value is now 0.23, which is slightly smaller than that of the translog model. The price ratio is negative and significant at a high threshold. As expected, an increase in the relative energy price relates to a decrease in energy intensity.³⁵ The software coefficient also shows a highly significant negative relationship. If the ratio between software and non-software capital increases by 1 percent, energy intensity decreases by 0.003 percent. Consequently, this effect size is slightly smaller but comparable to average elasticities derived with the translog model and is very similar to the Equation (4) elasticity at averages of S_E and S_{SW} . It should be emphasized here that the coefficient could be slightly biased since we assume a linear relationship. Moreover, we now measure the

 $^{^{34}\}mathrm{We}$ lose a small fraction of observations here as the producer price index is not available for the repair and installation industry (Division 33) for 2009. ³⁵Please note here that this relationship is not strictly exogenous.

effect of software capital relative to tangible capital and no longer the effect of software capital relative to output. In a second specification, we re-estimate the model but exclude potentially problematic observations with respect to "zero" software capital stocks. The software capital coefficient is now only significant at a 95%-level. We observe a similar phenomenon for the translog model. Furthermore, the magnitude of the effect increases, but this increase is not substantial. Therefore, we still consider the effect size as inelastic. In the third column, we only consider increasing software capital stocks. The coefficient of software capital relative to tangible capital is comparable to the first specification, however, it is only significant at a 99%-threshold. Moreover, the adjusted R^2 slightly increases. To sum up, results based on a reduced form of a CES-production function are robust to different econometric specifications and comparable to those derived with the translog model.

7. Discussion

Previous studies on manufacturing industries point out that the ongoing digital transformation may have synergies with climate targets. A higher amount of data and an improved exploitation of information increases efficiency within production processes and may decrease relative energy use, despite the fact that ICT consume energy themselves.

To the best of our knowledge, this is the first empirical study that uses firm-level data to analyze the validity of this claim. Using software capital intensity as a proxy for the firm-level degree of digitalization, we find that an increase thereof relates to a decrease in relative energy use, however, to a much smaller magnitude than previous sector-level estimates state. Consequently, the relationship is highly inelastic. Nevertheless, we would like to emphasize that we cannot fully rule out that estimated coefficients are small because they are downward biased due to a measurement error in software capital. However, the estimated software capital coefficient is robust to several sample restrictions and different modifications of software capital stocks. This consistency may provide some confidence.

It is not unusual that effects are smaller when microeconometric data is employed. In a meta analysis on the relationship between IT and productivity, Stiroh (2005) observes a similar phenomenon. The respective elasticity tends to be larger at the industry level and including firm-level fixed effects decreases the magnitude of the relationship. Also, Kaus et al. (2020) find lower effects of intangibles on output at the firm level than Niebel et al. (2017) at the aggregated level.

Furthermore, we analyze the robustness of results with respect to different firmlevel characteristics. For instance, for most industries we find that the effect size is not significant, but software-related reductions in energy costs are significant and tend to be more pronounced for exactly those firms and industries that are comparatively energy intensive. Thus, there is some ray of hope.

Moreover, although the relationship is small, it does not necessarily mean that it is not relevant as software capital grew strongly in our sample in the observed time frame. Hence, software capital may still relate to considerable energy intensity improvements due to its large growth rate. To analyze this, we perform the following back-of-theenvelope calculation. We multiply the demand elasticity by the average annual growth rate of software capital intensity, which is 18.05 percent. Using the translog model, this translates into an annual decrease in energy intensity between 0.12 and 0.20 percent, depending on the approach used to calculate the elasticity. Over ten years this would result in energy intensity improvements of roughly 1.5 percent, assuming that software capital would continue to grow at such a high rate. This shows that software investments do relate to energy intensity improvements to some extent, but are not a key driver for achieving climate targets.

A further question is whether it would be economically rational to invest in software to save energy. To shed light into economic considerations, we approximate average savings in energy costs per euro invested in software. We use the average energy intensity elasticity derived by Equation (3) in Column (1) of Table 3 to calculate energy cost savings per euro invested in software. In Appendix E, our approach is described in detail. We measure that $1 \in$ invested in software saved on average approximately $0.03 \in$ of energy costs in the analyzed time frame. This calculation illustrates that conducting software investments to save energy (costs) does generally not appear to be economical from a firm perspective. This confirms a rational already observed in questionnaire-based surveys, in which firms were asked for non-technical self-assessments: Savings in energy consumption due to the use of digital technologies are rather a welcome side effect and do not appear to be substantial enough to be the main motivation for conducting digitalization projects.

What does our result imply for the net impact of ICT on total energy consumption? As the level of output is considered in both econometric models, the measured relationship indicates not necessarily a decrease in absolute energy consumption. By means of the translog model, we analyze the relationship between software capital intensity and the ratio between energy and labor costs. By estimating a reduced form of a CES production function approach, we consider the relationship between software usage and the ratio between energy use and output. Many economic studies show a clear link between labor and ICT (e.g., Van Reenen 2011, Michaels et al. 2014 and Atasoy et al. 2016) as well as productivity and ICT (Stiroh, 2005; Cardona et al., 2013). In other words, the observed relationship may be exclusively driven by positive effects of software capital on labor and output. Accordingly, there is not necessarily a decrease in absolute energy use within firms. It may be just less affected by software usage. Therefore, we abstain to make conclusions on absolute energy consumption. Besides, we want to emphasize that even if output or labor increases due to software usage and energy consumption remains constant or grows to a lower extent, energy intensity improvements still occur, as energy is used relatively less. Moreover, to be precise, we solely measure energy intensity improvements inside firms. For instance, we cannot make conclusions about additional energy that is consumed in external data centers due to an increase in the use of Could Computing. However, Cloud Computing has been not used very frequently in the observed time frame and its use has only picked up in more recent years.³⁶ An analysis with more recent data would probably run into greater difficulties here.

One issue of this study may be that energy efficiency improvements may be accompanied by rebound effects (Amjadi et al., 2018). For example, potential energy savings may not be fully realized because improvements in energy efficiency increase the attractiveness of using energy as an input factor. How the use of ICT relates to this issue could be worth looking into in further research.

Furthermore, effects may be small because software capital is insufficient to approximate the degree of digitalization. Unlike other digitalization indicators, e.g., the amount of employees working with a computer, software capital has the advantage that it is measured in monetary values. In addition, almost all hardware requires software. Especially technologies that optimize production by analyzing large amounts of data and thus potentially improve energy efficiency, heavily rely on software. We show a clear relationship between software usage and the use of other digital technologies in Section

³⁶See website link (accessed 12. Dec 2021).

4.2. Consequently, another advantage in using software capital is that it is very general in comparison to other technologies such as Cloud Computing or 3D printing. Considering all possible indicators, we believe that for the purpose of this study, software capital is the most suitable indicator. Nonetheless, further analyses using different types of digital technologies may be useful, as heterogeneous effects with respect to different types of digital technologies could exist. Moreover, we have to acknowledge the issue that we do not consider the use of software that is free of charge. However, as we only look at relative percentage changes and it is likely that for most firms both, the use of free of charge and paid software, are proportional to each other, we assume that this does not have a large effect on our results.

Besides, we want to mention that AFiD data provide information on the use of different energy sources. Hence, it is also possible to analyze the influence of software capital on the energy mix, which may be affected by digital technologies, as well. However, a respective analysis would also require information on energy costs and prices for different sources. Unfortunately, the matched price data is often not sufficiently accurate, and different specifications of cost shares for each energy source yielded unreliable estimates. Therefore, we abstained from the analysis. This does not affect our main findings, as information on overall energy costs are surveyed by Destatis.

8. Conclusion and Outlook

Climate change and the emergence of digital technologies are considered as current megatrends. Consequently, to analyze how both relate to each other is of great importance. In particular, it is assumed that digital technologies may relate to a decrease in carbon emissions through energy intensity improvements especially in manufacturing industries, as they are considered to have very high potentials in this regard.

This is the first large-scale empirical study that analyzes the relationship between the usage of digital technologies and energy intensity improvements at the firm level. For this purpose, we employ administrative panel data on 28,600 firms in the German manufacturing sector collected between 2009 and 2017. Furthermore, we use software capital intensity as an indicator for the firm-level degree of digitalization and apply a translog cost function approach. Results show a statistically significant link between software capital and energy intensity improvements, but the effect size is much smaller than expected. Our findings are robust to several sample restrictions as well as to modifications of the software capital stock. Thus, we conclude that an increase in the firm-level software capital stock cannot be associated with substantial energy intensity improvements within firms. However, there is some ray of hope, as we find that effects are more pronounced in firms and industries that are very energy intensive.

Our results may be relevant for policy makers, consultants and firms that aim to improve energy intensity within establishments and may overestimate synergies between digital technologies and energy savings.

Moreover, effects are small for software usage, but they may be different for specific digital technologies. Future research that analyzes how different digital technologies relate to effect heterogeneity would be a further important contribution, for which the application of firm-level data has great potential. External factors such as carbon prices or market concentration may also incentivize the use of digital technologies to improve energy intensity and thus potentially influence effect heterogeneity. An analysis in this regard may be very useful for designing appropriate policies dealing with climate protection potentials of digital technologies. Last but not least, the analysis of ICT-related rebound effects as well as the inclusion of an appropriate instrumental variable that allows measuring whether the relationship is truly causal would be of great value.

Acknowledgments

We thank Irene Bertschek, Peter Winker, Ulrich Wagner, Jens Clausen, Ralph Hintemann and Grazia Cecere for valuable feedback as well as Kerstin Stockmeyer and Stefan Seitz for supporting access to the German administrative data. We are grateful for the seminar invitation and feedback received from participants of MINES ParisTech and Telecom Paris. This work also benefited from presentations at the Verein für Socialpolitik Jahrestagung 2021, the 48th EARIE Annual Conference in Bergen, the 26th Annual Conference of the European Association of Environmental and Resource Economists in Berlin, the ITS Biennial Conference 2021 in Gothenburg, the International Energy Workshop 2021 in Freiburg, the Digitalisation Research and Network Meeting - DigiMeet 2021, the 1st IAEE Online Conference and the 9th Mannheim Conference on Energy and the Environment as well as from seminars at ZEW Mannheim.

Funding

This paper has been written as part of the research project "CliDiTrans" Climate Protection Potential of Digital Transformation. The project received funding from the German Federal Ministry of Education and Research (funding ID: 01LA1818B), which played no role in the research.

References

Allison, P. D. (2009), Fixed Effects Regression Models, London: Sage.

- Amjadi, G., Lundgren, T. and Persson, L. (2018), 'The rebound effect in Swedish heavy industry', *Energy Economics* 71, 140–148.
- Andrae, A. S. G. and Edler, T. (2015), 'On global electricity usage of communication technology: Trends to 2030', *Challenges* 6(1), 117–157.
- Atasoy, H., Banker, R. D. and Pavlou, P. A. (2016), 'On the longitudinal effects of IT use on firm-level employment', *Information Systems Research* 27(1), 6–26.
- Belkhir, L. and Elmeligi, A. (2018), 'Assessing ICT global emissions footprint: Trends to 2040 & recommendations', Journal of Cleaner Production 177, 448–463.
- Berkhout, F. and Hertin, J. (2004), 'De-materialising and re-materialising: Digital technologies and the environment', *Futures* 36(8), 903–920.
- Berlemann, M. and Wesselhöft, J.-E. (2014), 'Estimating aggregate capital stocks using the perpetual inventory method', *Review of Economics* **65**(1), 1–34.
- Berndt, E. R. and Hesse, D. M. (1986), 'Measuring and assessing capacity utilization in the manufacturing sectors of nine OECD countries', *European Economic Review* 30(5), 961–989.
- Berndt, E. and Wood, D. (1975), 'Technology, prices, and the derived demand for energy', The Review of Economics and Statistics 57(3), 259–268.
- Bernstein, R. and Madlener, R. (2010), 'Impact of disaggregated ICT capital on electricity intensity in European manufacturing', Applied Economics Letters 17(17), 1691– 1695.
- Bertschek, I., Cerquera, D. and Klein, G. J. (2013), 'More bits more bucks? Measuring the impact of broadband internet on firm performance', *Information Economics and Policy* **25**(3), 190–203.
- Bertschek, I., Erdsiek, D., Niebel, T., Schuck, B., Seifried, M., Ewald, J., Lang, T., Hicking, J., Wenger, L. and Walter, T. (2020), *Schwerpunktstudie Digitalisierung* und Energieeffizienz - Erkenntnisse aus Forschung und Praxis, Bundesministerium für Wirtschaft und Energie (BMWi), Berlin.
- Brown, R. S. and Christensen, L. R. (1980), Estimating elasticities of substitution in a model of partial static equilibrium: An application to U.S. agriculture, 1947-1974, SSRI Workshop Series 292581, University of Wisconsin-Madison, Social Systems Research Institute. https://ideas.repec.org/p/ags/uwssri/292581.html.
- Brozzi, R., Forti, D., Rauch, E. and Matt, D. T. (2020), 'The advantages of industry 4.0 applications for sustainability: Results from a sample of manufacturing companies', *Sustainability* 12(9), 3647.
- Cardona, M., Kretschmer, T. and Strobel, T. (2013), 'ICT and productivity: Conclusions from the empirical literature', *Information Economics and Policy* 25(3), 109–125.
- Chen, X., Despeisse, M. and Johansson, B. (2020), 'Environmental sustainability of digitalization in manufacturing: A review', *Sustainability* **12**(24), 10298.

- Chimbo, B. et al. (2020), 'Information and communication technology and electricity consumption in transitional economies', *International Journal of Energy Economics and Policy* **10**(3), 296–302.
- Christensen, L. R., Jorgenson, D. W. and Lau, L. J. (1973), 'Transcendental logarithmic production frontiers', The Review of Economics and Statistics 55(1), 28–45.
- Collard, F., Fève, P. and Portier, F. (2005), 'Electricity consumption and ICT in the French service sector', *Energy Economics* **27**(3), 541–550.
- Corbet, S., Lucey, B. and Yarovaya, L. (2021), 'Bitcoin-energy markets interrelationships New evidence', *Resources Policy* **70**, 101916.
- Crépon, B. and Heckel, T. (2002), 'Computerization in France: An evaluation based on individual company data', *Review of Income and Wealth* 48(1), 77–98.
- Dhyne, E., Konings, J., den Bosch, J. V. and Vanormelingen, S. (2018), IT and productivity: A firm level analysis, NBB Working Paper 346, National Bank of Belgium. http://hdl.handle.net/10419/207726.
- Draca, M., Sadun, R. and Reenen, J. V. (2007), Productivity and ICTs: A review of the evidence, *in* 'The Oxford Handbook of Information and Communication Technologies', R. Mansell, C. Avgerou, D. Quah and R. Silverstone (eds.), Oxford University Press, pp. 100–147.
- European Commission (2019), 'COM(2019) 640 final: The European Green Deal'. https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=COM:2019:640:FIN.
- European Parliament and the Council of the European Union (2018), 'PE/55/2018/REV/1. Regulation (EU) 2018/1999 of the European Parliament and of the Council of 11 December 2018 on the Governance of the Energy Union and Climate Action, amending Regulations (EC) No 663/2009 and (EC) No 715/2009 of the European Parliament and of the Council, Directives 94/22/EC, 98/70/EC, 2009/31/EC, 2009/73/EC, 2010/31/EU, 2012/27/EU and 2013/30/EU of the European Parliament and of the Council Directives 2009/119/EC and (EU) 2015/652 and repealing Regulation (EU) No 525/2013 of the European Parliament and of the Council (Text with EEA relevance.)'. https://eur-lex.europa.eu/eli/reg/2018/1999/oj.
- European Union (2019), 'Masterplan for a competitive transformation of EU energyintensive industries enabling a climate-neutral, circular economy by 2050.', *Publications Office of the European Union*. https://op.europa.eu/s/u9wx [Online; accessed 12. Nov. 2021].
- European Union (2021), 'Survey on the contribution of ICT to the environmental sustainability actions of EU enterprises'. https://digital-strategy.ec.europa.eu/ en/library/survey-contribution-ict-environmental-sustainability-action s-eu-enterprises [Online; accessed 12. Nov. 2021].
- Ferreboeuf, H., Berthoud, F., Bihouix, P., Fabre, P., Kaplan, D. and Lefèvre, L. (2019), Lean ICT: Towards Digital sobriety, The Shift Project. https://theshiftproject. org/en/article/lean-ict-our-new-report/.
- Foster-McGregor, N., Stehrer, R. and de Vries, G. J. (2013), 'Offshoring and the skill structure of labour demand', *Review of World Economics* 149(4), 631–662.

- Friedrich, R., Ploner, F., Schäfer, C. T., Disselhoff, T., Petkau, A., Hennemann, C., Moecke, J., Wätzig, T., Zimmert, O., Waltersmann, L., Kiemel, S., Miehe, R. and Sauer, A. (2021), 'Potenziale der schwachen künstlichen Intelligenz für die betriebliche Ressourceneffizienz'. https://www.ressource-deutschland.de/publikationen/s tudien/ [Online; accessed 12. Nov. 2021].
- Frisch, R. and Waugh, F. V. (1933), 'Partial time regressions as compared with individual trends', *Econometrica: Journal of the Econometric Society* pp. 387–401.
- German Environment Agency (2021), 'Energieverbrauch nach Energieträgern und Sektoren'. https://www.umweltbundesamt.de/daten/energie/energieverbrauch-na ch-energietraegern-sektoren [Online; accessed 20. Nov 2021].
- GeSI & Accenture (2015), 'Smarter 2030 ICT solutions for 21st century challenges'. http://smarter2030.gesi.org/downloads/Full_report.pdf [Online; accessed 20. April 2021].
- Ghobakhloo, M. and Fathi, M. (2021), 'Industry 4.0 and opportunities for energy sustainability', Journal of Cleaner Production 295, 126427.
- Griliches, Z. (1980), 'R&D and the productivity slowdown', National Bureau of Economic Research - Working Paper Series . https://www.nber.org/papers/w0434.
- Haller, S. A. and Hyland, M. (2014), 'Capital-energy substitution: Evidence from a panel of Irish manufacturing firms', *Energy Economics* 45, 501–510.
- Hijzen, A., Görg, H. and Hine, R. C. (2005), 'International outsourcing and the skill structure of labour demand in the united kingdom', *The Economic Journal* 115(506), 860– 878.
- Hilty, L. M., Arnfalk, P., Erdmann, L., Goodman, J., Lehmann, M. and Wäger, P. A. (2006), 'The relevance of information and communication technologies for environmental sustainability – A prospective simulation study', *Environmental Modelling & Software* 21(11), 1618–1629.
- IEA (2018), 'Digitalization and Energy Analysis IEA'. https://www.iea.org/repo rts/digitalisation-and-energy [Online; accessed 25. Nov. 2020].
- IEA (2020a), 'Energy efficiency indicators: Overview'. https://www.iea.org/report s/energy-efficiency-indicators-overview [Online; accessed 29. Oct. 2021].
- IEA (2020b), 'Tracking Industry 2020 Analysis IEA'. https://www.iea.org/report s/tracking-industry-2020 [Online; accessed 19. Nov. 2020].
- IEA (2021), 'Global Energy Review 2021 Analysis IEA'. https://www.iea.org/ reports/global-energy-review-2021/co2-emissions [Online; accessed 13. Oct. 2021].
- IPCC (2018), Summary for Policymakers., in 'Global Warming of 1.5°C. An IPCC Special Report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty', Masson-Delmotte, V., P. Zhai, H.-O. Pörtner, D. Roberts, J. Skea, P.R. Shukla, A. Pirani, W. Moufouma-Okia, C. Péan, R. Pidcock, S. Connors, J.B.R. Matthews, Y. Chen, X. Zhou, M.I. Gomis, E. Lonnoy, T. Maycock, M. Tignor, and T.

Waterfield (eds.), World Meteorological Organization, Geneva, Switzerland, p. 32 pp. https://www.ipcc.ch/site/assets/uploads/2018/10/SR15_SPM_version_stand _alone_LR.pdf.

- Kaus, W., Slavtchev, V. and Zimmermann, M. (2020), Intangible capital and productivity: Firm-level evidence from German manufacturing, IWH Discussion Papers 1/2020, Leibniz-Institut für Wirtschaftsforschung Halle (IWH). http://hdl.handle.net/104 19/213561.
- Kratena, K. (2007), 'Technical change, investment and energy intensity', Economic Systems Research 19(3), 295–314.
- Kusiak, A. (2018), 'Smart manufacturing', International Journal of Production Research 56(1-2), 508–517.
- Lagomarsino, E. (2020), 'Estimating elasticities of substitution with nested CES production functions: Where do we stand?', *Energy Economics* 88, 104752.
- Lahouel, B. B., Taleb, L., Zaied, Y. B. and Managi, S. (2021), 'Does ICT change the relationship between total factor productivity and CO2 emissions? Evidence based on a nonlinear model', *Energy Economics* 101, 105406.
- Lange, S., Pohl, J. and Santarius, T. (2020), 'Digitalization and energy consumption. Does ICT reduce energy demand?', *Ecological Economics* 176, 106760.
- Löschel, A., Lutz, B. J. and Managi, S. (2019), 'The impacts of the EU ETS on efficiency and economic performance – An empirical analyses for German manufacturing firms', *Resource and Energy Economics* 56, 71–95.
- Lutz, B. J. (2016), Emissions trading and productivity: Firm-level evidence from German manufacturing, ZEW Discussion Papers 16-067, ZEW Leibniz Centre for European Economic Research. https://ideas.repec.org/p/zbw/zewdip/16067.html.
- Malmodin, J. and Lundén, D. (2018), 'The energy and carbon footprint of the global ICT and E&M sectors 2010–2015', Sustainability 10(9), 3027.
- Masanet, E., Shehabi, A., Lei, N., Smith, S. and Koomey, J. (2020), 'Recalibrating global data center energy-use estimates', *Science* 367(6481), 984–986.
- Michaels, G., Natraj, A. and van Reenen, J. (2014), 'Has ICT polarized skill demand? Evidence from eleven countries over twenty-five years', *The Review of Economics and Statistics* 96(1), 60–77.
- Mundlak, Y. (1978), 'On the Pooling of Time Series and Cross Section Data', Econometrica 46(1), 69–85.
- Niebel, T., O'Mahony, M. and Saam, M. (2017), 'The contribution of intangible assets to sectoral productivity growth in the EU', *Review of Income and Wealth* 63, 49–67.
- OECD (2017), The Next Production Revolution: Implications for Governments and Business, OECD Publishing, Paris. https://www.oecd-ilibrary.org/content/public ation/9789264271036-en.
- Paris Agreement (2015), 'United Nations Treaty Collection'. https://treaties.un.or g/doc/Treaties/2016/02/20160215%2006-03%20PM/Ch_XXVII-7-d.pdf [Online; accessed 15. Oct. 2021].

- Patterson, M. G. (1996), 'What is energy efficiency? concepts, indicators and methodological issues', *Energy Policy* 24(5), 377–390.
- Preist, C., Schien, D. and Shabajee, P. (2019), Evaluating sustainable interaction design of digital services: The case of YouTube, *in* 'Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems', pp. 1–12.
- Santarius, T., Pohl, J. and Lange, S. (2020), 'Digitalization and the decoupling debate: Can ICT help to reduce environmental impacts while the economy keeps growing?', *Sustainability* 12(18), 7496.
- Schulte, P., Welsch, H. and Rexhäuser, S. (2016), 'ICT and the demand for energy: Evidence from OECD countries', *Environmental and Resource Economics* 63(1), 119– 146.
- Stiroh, K. J. (2005), 'Reassessing the impact of IT in the production function: A metaanalysis and sensitivity tests', Annales d'Économie et de Statistique (79/80), 529–561.
- Stoll, C., Klaaßen, L. and Gallersdörfer, U. (2019), 'The carbon footprint of Bitcoin', Joule 3(7), 1647–1661.
- Thoben, K.-D., Wiesner, S. A. and Wuest, T. (2017), "Industy 4.0" and smart manufacturing – a review of research issues and application examples', *International Journal* of Automation Technology **11**(1), 4–19.
- Van der Werf, E. (2008), 'Production functions for climate policy modeling: An empirical analysis', *Energy Economics* **30**(6), 2964–2979.
- Van Heddeghem, W., Lambert, S., Lannoo, B., Colle, D., Pickavet, M. and Demeester, P. (2014), 'Trends in worldwide ICT electricity consumption from 2007 to 2012', Computer Communications 50, 64–76.
- Van Reenen, J. (2011), 'Wage inequality, technology and trade: 21st century evidence', Labour Economics 18(6), 730–741.
- Walker, W. (1985), 'Information technology and the use of energy', *Energy Policy* 13(5), 458–476.
- Weber, T., Bertschek, I., Ohnemus, J. and Ebert, M. (2018), 'Monitoring-Report Wirtschaft DIGITAL 2018'.
- Wen, H., Lee, C.-C. and Song, Z. (2021), 'Digitalization and environment: How does ICT affect enterprise environmental performance?', *Environmental Science and Pollution Research* 28(39), 54826–54841.
- Zhang, W., Gu, F. and Guo, J.-F. (2019), 'Can smart factories bring environmental benefits to their products? A case study of household refrigerators', *Journal of Industrial Ecology* 23(6), 1381–1395.

Appendix A. Additional Data

For our analysis, we add, inter alia, information on prices of different energy sources, gross value added deflators to calculate real value added and growth and depreciation rates as well as investment deflators to calculate capital stocks. All data sources are listed in Table A.5. The identifier denotes the variable that is used to merge the dataset with AFiD.

Table A.5: Description of additional data sources.

Information	Data source	Comments	Identifier
Price for energy source (electricity, natural gas, heating oil, coal)	Gesamtausgabe der En- ergiedaten, Federal Ministry for Economic Affairs and Energy (BMWi), status: 31.03.2020, link to website (Retrieved on: 01.04.2020)	Prices for hard coal (import prices), heavy heating oil (in- dustry prices, VAT excluded), light heating oil (light, industry prices, VAT excluded), electric- ity and natural gas prices inde- pendent from the consumption level. are retrieved. The respec- tive units have all been converted to \notin /kWh.	Year
Price for energy source (district heat)	Fernwärme – Preisübersicht, AGFW Der Energieeffizien- zverband für Wärme, Kälte und KWK e. V., status: 01.10.2017, link to website (Retrieved on: 14.08.2019)	Absolute price development from 2009-2017 for the connected loads of 160 kW (p.8) are used. Values are converted from ϵ /MWh to ϵ /kWh. Prices are retrieved without VAT.	Year
Price for energy source (biomass)	Brennstoffkostenentwicklung von Gas, Öl und Pellets, Deutsches Pelletinstitut GmbH (DEPI), status: 2019, link to website (Betrieved on: 13.09.2019)	Pellet price for 2015 is taken, value converted from cent/kWh to ϵ /kWh (VAT excluded).	Year
Price for energy source (biomass)	(netrieved off. 15.52.2019) Index der Erzeugerpreise gewerblicher Produkte (5.10 Holzprodukte - GP09-1629 14 908 Pellets, Briketts, Scheiten o.ä. Formen aus Sägespänen u.a. Sägenebenprodukt), from: Daten zur Energiepreisentwick- lung - Lange Reihen von Januar 2005 bis Mai 2020, Statistis- ches Bundesamt (Destatis), status: 26.06.2020, link to data (Detatis - 10.07200)	The base year of the Destatis index is 2015. Therefore, the DEPI-price is taken from the year 2015 and multiplied by the index for each year to receive informa- tion about the change in the price for biomass.	Year
Price for energy source (liquid gas)	(Retrieved on: 16.07.2020) IEA Energy Prices and Taxes Statistics, International Energy Agency, status: 1.Quarter 2019, link to website (Retrieved on: 04.09.2019)	Prices (VAT excluded) from 2009-2017 for liquid gas are retrieved. Values are converted from \notin /l to \notin /kWh.	Year
Producer price index (PPI)	Index der Erzeugerpreise gewerblicher Produkte (In- landsabsatz) nach dem Güter- verzeichnis für Produktion- sstatistiken Ausgabe 2009 (GP 2009) - Lange Reihen der Fach- serie 17, Reihe 2 von Januar 2005 bis September 2020, Statis- tisches Bundesamt (Destatis), status: 20.10.2020, link to data (Retrieved on: 12.11.2020)	Index on the yearly average change is retrieved.	Year, economic sectors (two-digit NACE code)
Gross value added deflators	National accounts aggregates by industry, Eurostat, status: 24.03.2020, Eurostat bookmark (Retrieved on: 01.04.2020)	Price index (implicit deflator), base year 2010, national cur- rency.	Year
Capital stock	Cross-classification of gross fixed capital formation by industry and by asset (flows) - Computer soft- ware and databases (gross), Eu- rostat, status: 30.03.2020, Eu- rostat bookmark (Retrieved on: 01.04.2020)	Table PD10_NAC, price index (implicit deflator), base year 2010, national currency. Soft- ware deflators are retrieved. See Appendix D for detailed informa- tion on how we calculate soft- ware as well as non-software cap- ital stocks.	Year

Information	Data source	Comments	Identifier
Capital stock	EU KLEMS database - 2019 release, Germany capital input data, see Stehrer, R., A. Bykova, K. Jäger, O. Reiter and M. Schwarzhappel (2019): Indus- try level growth and productivity data with special focus on intan- gible assets, wiiw Statistical Re- port No. 8. link to data (Re- trieved on: 18.04.2020)	Real gross fixed capital forma- tion (in prices from 2010) to cal- culate growth rates, depreciation rates as well as investment de- flators (except software deflators) are taken from the EU KLEMS database for the years 2003-2017. See Appendix D for detailed in- formation on how we calculate software as well as non-software capital stocks	Year, economic sectors (two-digit NACE code)
Household broadband availability	Breitbandatlas des Bundes (German Broadband Atlas) - Release 2/2021 (Retrieved on: 9.04.2021). Data is restricted in usage. Access can be requested at atene KOM GmbH (link to website)	Not integrated in the analysis	municipality level (AGS)

Category	Summarized energy carriers					
Biomass	Solid biogenic substances, liquid biogenic substances, biogas,					
	sewage gas, landfill gas, sewage sludge					
Natural gas	Natural gas, petroleum gas					
Coal	Hard coals, hard coal coke, raw lignites, lignite briquettes, hard					
	coal briquettes, other hard coals, lignite cokes, fluidized bed coals,					
	pulverized and dry coals, other lignite					
Heating oil	Light and heavy heating oil					
District heat	District heat					
Liquid gas	Liquid gas					
Other energy sources	Mine gas, coke oven gas, blast furnace gas, converter gas, other					
	gases, waste (household waste, industrial waste), other energy car-					
	riers (waste heat, etc.)					

Appendix B. Categorization of Different Energy Carriers

Table B.6: Categorization of different energy carriers.

Appendix C. Derivation of Schulte et al.'s (2016) Dual Cost Function Model

Variable costs are defined by energy (E) and labor (L) use and the corresponding energy (P_E) and labor prices (P_L) .

$$VC = P_E E + P_L L \tag{C.1}$$

Moreover, the restricted variable cost function depends on the following parameters, which are defined in Section 3.

$$VC = f(P_E, P_L, K_{ICT}, K_N, Y, t)$$
(C.2)

This relationship is approximated by a translog cost function:

$$\begin{aligned} \ln VC = \beta_0 + \beta_E \ln P_E + \beta_L \ln P_L + \beta_{K_{ICT}} \ln K_{ICT} + \beta_{K_N} \ln K_N + \beta_Y \ln Y + \beta_T t \\ + \frac{1}{2} \beta_{EE} \ln (P_E)^2 + \frac{1}{2} \beta_{LL} \ln (P_L)^2 + \frac{1}{2} \beta_{K_{ICT}K_{ICT}} \ln (K_{ICT})^2 \\ + \frac{1}{2} \beta_{EL} \ln (K_N)^2 + \frac{1}{2} \beta_{YY} \ln (Y)^2 + \frac{1}{2} \beta_{TT} (t)^2 \\ + \frac{1}{2} \beta_{EL} \ln P_E \ln P_L + \beta_{EK_{ICT}} \ln P_E \ln K_{ICT} + \beta_{EK_N} \ln P_E \ln K_N \\ + \beta_{EY} \ln P_E \ln Y + \delta_{ET} \ln P_E t \\ + \frac{1}{2} \beta_{LE} \ln P_L \ln P_E + \beta_{LK_{ICT}} \ln P_L \ln K_{ICT} + \beta_{LK_N} \ln P_L \ln K_N \\ + \beta_{LY} \ln P_L \ln Y + + \delta_{LT} \ln P_L t \\ + \frac{1}{2} \beta_{K_{ICT}K_N} \ln K_{ICT} \ln K_N + \beta_{K_{ICTY}} \ln K_{ICT} \ln Y + \delta_{K_{ICT}T} \ln K_{ICT} t \\ + \frac{1}{2} \beta_{K_N K_{ICT}} \ln K_N \ln K_{ICT} + \beta_{K_{NY}} \ln K_N \ln Y + \delta_{K_N T} \ln K_N t \\ + \delta_{YT} \ln Y t \end{aligned}$$

$$(C.3)$$

Applying logarithmic differentiation with respect to the energy price and Shephard's lemma, leads to Equation (C.4).

$$\frac{\partial lnVC}{\partial lnP_E} = \frac{P_E E}{VC} = S_E = \beta_E + \frac{1}{2}\beta_{EL}lnP_L + \frac{1}{2}\beta_{LE}lnP_L + \beta_{EE}lnP_E + \beta_{EK_N}lnK_N + \beta_{EK_{ICT}}lnK_{ICT} + \beta_{EY}lnY + \delta_{ET}t$$
(C.4)

Assuming symmetry ($\beta_{EL} = \beta_{LE}$) and homogeneity of degree one ($\beta_{EL} = -\beta_{EE}$) (see Christensen et al. 1973 and Berndt and Wood 1975) enables the transformation to the estimation equation $S_E = \beta_E + \beta_{EE} ln P_E / P_L + \beta_{EK_N} ln K_N + \beta_{EK_{ICT}} ln K_{ICT} + \beta_{EY}^* ln Y + \delta_{ET} t.^{37}$

The demand elasticity is derived following Kratena (2007). The demand elasticity of a good j can be defined as the change in $lnj \in \{E, L\}$ with respect to lnK_{ICT} . Expressing j as S_jVC/P_j allows decomposing the demand elasticity into three different effects. The effect of ICT on the share of energy costs in variable, the effect of ICT on total variable costs and the effect of ICT on prices.

³⁷With $\beta_{EY}^* = \beta_{EY} + \beta_{EK_N} + \beta_{EK_{ICT}}$

$$\epsilon_{jK_{ICT}} = \frac{\partial \ln j}{\partial \ln K_{ICT}} = \frac{\partial \ln \frac{S_j V C}{P_j}}{\partial \ln K_{ICT}} = \frac{\partial \ln S_j}{\partial \ln K_{ICT}} + \frac{\partial \ln V C}{\partial \ln K_{ICT}} - \frac{\partial \ln P_j}{\partial \ln K_{ICT}}$$
(C.5)

Assuming exogenous prices implies $\partial \ln P_j / \partial \ln K_{ICT} = 0$, which leads to Equation (C.6).

$$\epsilon_{jK_{ICT}} = \frac{\partial \ln S_j}{\partial \ln K_{ICT}} + \frac{\partial \ln VC}{\partial \ln K_{ICT}} \tag{C.6}$$

Which can be also expressed as:

$$\epsilon_{jK_{ICT}} = \frac{\beta_{jK_{ICT}}}{S_j} + \frac{\partial VC}{\partial K_{ICT}} \frac{K_{ICT}}{VC} \tag{C.7}$$

Assuming that $\partial VC/\partial K_{ICT}$ is a shadow price for capital allows writing Equation (C.8).

$$\epsilon_{jK_{ICT}} = \frac{\beta_{jK_{ICT}}}{S_j} - \frac{R_{K_{ICT}}K_{ICT}}{VC} \tag{C.8}$$

Furthermore, according to Schulte et al. (2016) $(R_{K_{ICT}}K_{ICT})/VC$ can be approximated by the share of ICT capital cost to variable costs $(S_{K_{ICT}})$. We assume a shadow price of software capital of $0.4 \in$.

$$\epsilon_{jK_{ICT}} = \frac{\beta_{jK_{ICT}}}{S_j} - S_{K_{ICT}} \tag{C.9}$$

Appendix D. Perpetual Inventory Method (PIM)

In the spirit of Griliches (1980), Berlemann and Wesselhöft (2014), Lutz (2016), Dhyne et al. (2018) and Löschel et al. (2019) capital stocks are calculated for software capital and non-software capital by means of the perpetual inventory method (PIM).

Given geometric constant depreciation, the capital stock K_t at period t can be written as a function of previous period's capital stock K_{t-1} , gross investments I_t , and the consumption of fixed capital at rate δ . Hence, capital stocks except initial ones can be calculated by the following equation.

$$K_t = (1 - \delta)K_{t-1} + I_t \tag{D.1}$$

To calculate initial capital stocks, one can express annual percentage increase in capital as the amount of investments minus the capital depreciated in the previous period.

$$\frac{K_t - K_{t-1}}{K_{t-1}} = \frac{I_t}{K_{t-1}} - \delta$$
(D.2)

Assuming that capital grows at a constant rate $(g_K = (K_t - K_{t-1})/K_{t-1})$, one can obtain the following expression.

$$K_{t-1} = \frac{I_t}{g_K + \delta} \tag{D.3}$$

Setting t = 1 allows to calculate the initial capital stock.

$$K_0 = \frac{I_1}{g_K + \delta} \tag{D.4}$$

For the calculation of firm-level initial capital stocks, it is recommended to use average investments of the first three years within the observation period because investments highly fluctuate over time.³⁸

$$\hat{I}_1 = \frac{\sum_{t=1}^3 I_t}{n}$$
(D.5)

Accordingly, in this study we calculate initial capital stocks by applying Equation (D.4) and (D.5), subsequent capital stocks are calculated by Equation (D.1).

PIM requires information on capital growth rates. These are estimated by calculating the compound annual growth rate at industry level using real gross fixed capital formation at prices from 2010. Information on gross fixed capital formation volume of software and total capital is retrieved from the EU KLEMS database. Depreciation rates and deflators for non-software capital are also taken from the EU KLEMS database. Software capital deflators are retrieved from Eurostat (see Table A.5).

 $^{^{38}}$ Please note here that we do robustness checks with respect to different period lengths to calculate initial capital stocks.

Appendix E. Calculation of Energy Cost Savings per Software Investment

By Equation (E.1), we initially calculate relative improvements in energy intensity per year related to software usage (*relative savings*). To do this, we multiply the energy intensity elasticity ($\epsilon_{EK_{SW}}$) by the relative change in software capital (ΔlnK_{ICT}) for each firm *i* in year *t*.

$$relative \ savings_{it} = \epsilon_{EK_{SW_{it}}} \times \Delta lnK_{SW_{it}} \tag{E.1}$$

To calculate savings in energy consumption, we assume that output is constant and calculate how much energy consumed in the previous period has been saved in the current period with respect to changes in the software capital stock (Equation E.2). Savings in energy costs are then approximated by multiplying savings in energy consumption by the firm-specific energy price.

$$cost \ savings_{it} = relative \ savings_{it} \times E_{i,t-1} \times P_E \tag{E.2}$$

In order to estimate the average savings in energy costs per euro invested in software, we sum up firm-level energy cost savings over all periods for which we have information and divide them by the sum of all software investments that have taken place in the same time period (Equation E.3).

$$savings \ per \ investment = \frac{\sum_{t=2010}^{2017} absolute \ savings_{it}}{\sum_{t=2010}^{2017} software \ investments_{it}}$$
(E.3)

Appendix F. Additional Descriptive Statistics

Appendix F.1. Distribution of Energy Prices



Figure F.6: Distribution of P_E .

Figure F.7: Distribution of P_E [external].



Figure F.8: Relationship between both energy prices.





Figure F.9: Average software capital intensity by region between 2009–2017. The dark blue regions represent those with the highest average software capital intensity. Regions with less than three observations per year or with no observations are not displayed.

Appendix F.3. Percentage change of mean (non-) software capital, labor and energy use divided by output (base year 2009)



Figure F.10: Change of mean (non-) software capital, labor and energy use divided by output (base year 2009).

Appendix F.4. Details on the Distribution of S_{SW} and S_E

	mean	sd	p5	p50	p95
$S_{K_{SW}}$	0.004	0.010	0.000	0.001	0.017
S_E	0.090	0.102	0.012	0.055	0.292
Observations	123362				

Table F.7: Detailed descriptive statistics on the distribution of S_{SW} and S_E .

Appendix G. Further Estimation Results

	(1)	(2)	(3)
		depreciation rates	
	25 percent	33 percent	50 percent
	ΔS_E	ΔS_E	ΔS_E
$\Delta \ln(\frac{P_E}{P_L})$	0.0284^{***}	0.0284^{***}	0.0284^{***}
	(61.74)	(61.74)	(61.74)
$\Delta \ln(\frac{K_{SW}}{Y})$	-0.000219***	-0.000237***	-0.000242***
	(-5.13)	(-5.18)	(-5.16)
$\Delta \ln(\frac{K_N}{Y})$	-0.0015***	-0.0015***	-0.0015***
	(-3.76)	(-3.77)	(-3.78)
$\Delta \ln(Y)$	0.0014^{*}	0.0013^{*}	0.0013^{*}
	(2.56)	(2.51)	(2.49)
Year	х	х	x
Economic sector	x	х	x
Multi-unit	x	х	x
Federal state	x	х	x
Size class	x	х	x
EEG exemption	х	х	x
Producer	х	х	x
Observations	89653	89653	89653
Adjusted \mathbb{R}^2	0.271	0.271	0.271

t statistics in parentheses. First-difference estimation. Clustered standard errors. + p<0.10, * p<0.05, ** p<0.01, *** p<0.001

Table G.8: Equation (5) with software capital stocks modified by different depreciation rates.

	(1)	(2)	(3)
	number of periods m	aximal included for initial so calculation	oftware capital
	2	4	6
	ΔS_E	ΔS_E	ΔS_E
$\Delta \ln(\frac{P_E}{P_L})$	0.0284^{***}	0.0284^{***}	0.0284^{***}
	(61.74)	(61.74)	(61.74)
$\Delta \ln(\frac{K_{SW}}{Y})$	-0.000167***	-0.000212***	-0.000325***
	(-4.68)	(-4.25)	(-4.69)
$\Delta \ln(\frac{K_N}{Y})$	-0.0015***	-0.0015***	-0.0015^{***}
	(-3.83)	(-3.82)	(-3.82)
$ riangle \ln(Y)$	0.0014^{**}	0.0013^{*}	0.0012^{*}
	(2.61)	(2.52)	(2.29)
Year	x	x	х
Economic sector	x	х	х
Multi-unit	x	x	х
Federal state	x	x	х
Size class	x	х	х
EEG exemption	x	x	х
Producer	х	х	х
Observations	89653	89653	89653
Adjusted \mathbb{R}^2	0.271	0.271	0.271

t statistics in parentheses. First-difference estimation. Clustered standard errors. + $p<0.10,\,^*~p<0.05,\,^{**}~p<0.01,\,^{***}~p<0.001$

Table G.9: Equation (5) with software capital stocks modified by different lengths of periods considered for the initial capital stock calculation.

	<i>4</i> .5			<i></i>
	(1)	(2)	(3)	(4)
	excluding potentially problematic observations	"zero" growth rates imputed	growth rates starting from "zero" imputed (I)	growth rates starting from "zero" imputed (II)
	ΔS_E	ΔS_E	ΔS_E	ΔS_E
$\Delta \ln(\frac{P_E}{P_L})$	0.0270***	0.0284***	0.0270***	0.0284***
	(50.15)	(61.74)	(51.13)	(61.74)
$\Delta \ln(\frac{K_{SW}}{Y})$	-0.000225^+	-0.000238***	-0.000356***	-0.000350***
	(-1.87)	(-5.14)	(-4.86)	(-4.77)
$\Delta \ln(\frac{K_N}{Y})$	-0.0017^{***}	-0.0015***	-0.0015^{**}	-0.0015***
	(-3.42)	(-3.78)	(-3.17)	(-3.76)
$ riangle \ln(Y)$	0.0011^{+}	0.0013^{*}	0.0011^{+}	0.0012^{*}
	(1.70)	(2.52)	(1.69)	(2.32)
Year	x	x	x	х
Economic sector	x	x	x	х
Multi-unit	x	x	x	х
Federal state	x	x	x	х
Size class	x	x	x	х
EEG exemption	x	x	x	х
Producer	x	x	x	x
N	65226	89653	66841	89653
adj. R^2	0.258	0.271	0.260	0.271

t statistics in parentheses. First-difference estimation. Clustered standard errors. + p<0.10, * p<0.05, ** p<0.01, *** p<0.001

Table G.10: Robustness checks with respect to "zero" software capital stocks.

	(1)	(2)	(3)	(4)	(5)
	after 2011	single-unit firms	no tangible capital	$\underset{K_{SW}}{\text{lagged}}$	sector-year fixed effects
	ΔS_E	ΔS_E	ΔS_E	ΔS_E	ΔS_E
$\Delta \ln(\frac{P_E}{P_L})$	0.0292***	0.0290^{***}	0.0284^{***}	0.0277^{***}	0.0284^{***}
	(52.32)	(59.68)	(61.74)	(45.84)	(61.80)
$\Delta \ln(\frac{K_{SW}}{Y})$	-0.000209***	-0.000177^{***}	-0.000249***		-0.000228***
	(-4.52)	(-3.78)	(-5.37)		(-4.95)
$\Delta \ln(\frac{K_N}{Y})$	-0.0013***	-0.0015^{***}		-0.0015^{**}	-0.0014^{***}
	(-3.30)	(-3.81)		(-3.13)	(-3.60)
$ riangle \ln(Y)$	0.00087	0.0015^{**}	0.0028^{***}	0.0000	0.0018^{***}
	(1.39)	(2.71)	(7.43)	(0.14)	(3.37)
$\Delta \ln(\frac{K_{SW}}{Y})_{t-1}$				0.0000562	
				(0.94)	
Year	x	x	x	x	x
Economic sector	x	x	x	x	x
Multi-unit	x	x	x	x	x
Federal state	x	x	x	x	x
Size class	x	x	x	x	x
EEG exemption	x	x	x	x	x
Producer	x	x	x	x	x
Economic sector \times Year					x
Observations	62821	77029	89653	59650	89653
adj. R^2	0.285	0.284	0.271	0.255	0.281

t statistics in parentheses. First-difference estimation. Clustered standard errors. * p<0.05, ** p<0.01, *** p<0.001

Table G.11: Further robustness checks (Equation (5)).

	(1)	(2)	(3)
	OLS	FE	Mundlack
	S_E	S_E	S_E
$ln(\frac{P_E}{P_L})$	-0.0038***	0.0316***	
	(-8.73)	(57.23)	
$\Delta ln(\frac{P_E}{R})$	× ,		0.0316^{***}
			(160.60)
$\mu(ln(\frac{P_E}{R}))$			-0.0054***
$P((P_L))$			(-8.06)
$ln(\frac{K_{SW}}{K_{SW}})$	-0.00165***	-0.000213***	(0.00)
V Y	(-32.95)	(-3.48)	
$\wedge ln(\frac{K_{SW}}{K_{SW}})$	(02.00)	(0.10)	-0.000212***
			(-5.77)
$u(ln(\frac{K_{SW}}{K_{SW}}))$			-0.00151***
$\mu(m(Y))$			(-17.63)
$ln(\frac{K_N}{N})$	0.0061***	-0.0001	(-11.00)
m(Y)	(30.85)	(-0.26)	
$\wedge ln(\frac{K_N}{K_N})$	(00.00)	(0.20)	-0.0001
			(-0.37)
$u(ln(\frac{K_N}{N}))$			0.0040***
$\mu(m(Y))$			(16.42)
ln(Y)	0.0397^{***}	0.0040***	(10.12)
	(69.70)	(6.18)	
$\triangle ln(Y)$	()	()	0.0040^{***}
			(11.96)
$\mu(ln(Y))$			0.0359***
			(48.22)
Year	х	х	x
Economic sector	х	х	х
Multi-unit	х	х	х
Federal state	х	х	х
Size class	х	х	х
EEG exemption	х	х	х
Producer	х	х	х
Observations 123362	123362	123362	
R^2	0.577	0.268	0.266

t statistics in parentheses. First difference estimation. Clustered standard errors except for Mundlak specification. p<0.05, ** p<0.01, *** p<0.001

Table G.12: Pooled OLS, fixed effects as well as Mundlak

	(1)	(2)	(3)	(4)
	capital	Y instru-	$\frac{P_E}{P_L}$	$Y, \frac{P_E}{P_L}$
	not	mented	instru-	instru-
	scaled by		mented	mented
	output			
	ΔS_E	ΔS_E	ΔS_E	ΔS_E
$\Delta \ln(\frac{P_E}{P_L})$	0.0284^{***}	0.0283^{***}	0.0023^{*}	0.0018^{+}
	(0.000)	(0.000)	(0.019)	(0.068)
$ riangle \ln(K_{SW})$	-0.000242^{***}	-0.000260***	-0.000245^{***}	-0.000265^{***}
	(0.000)	(0.000)	(0.000)	(0.000)
$ riangle \ln(K_N)$	-0.0016***	-0.0019***	-0.0012^{**}	-0.0015^{**}
	(0.000)	(0.000)	(0.006)	(0.002)
$ riangle \ln(Y)$	0.0031^{***}	0.011^{***}	0.0036***	0.0108^{**}
	(0.000)	(0.001)	(0.000)	(0.001)
Year	х	x	x	x
Economic sector	х	x	x	x
Multi-unit	x	x	x	x
Federal state	x	x	x	x
Size class	x	x	x	x
EEG exemption	x	x	x	x
Producer	x	x	x	x
Observations	89653	89017	89653	89017
Underidentification		64.16	816.0	63.83
Weak identification		42.61	1139.2	27.93
P-value Hansen J statistic		0.864		0.909

p-values in parentheses. First-difference estimation. Clustered standard errors.⁺ $p < 0.10, \ ^* p < 0.05, \ ^** p < 0.01, \ ^*** p < 0.001$. Under
identification test displays the Kleibergen-Paap LM statistic and the weak identification test displays the Kleibergen-Paap Wald F-statistic.

Table G.13: IV estimates

Appendix H. Results – Firm-level Characteristics



Figure H.11: Heterogeneous effects - Energy and software capital intensity (1st Quartile: Lowest level of average energy or software capital intensity)



Figure H.12: Heterogeneous effects - Time differences



Figure H.13: Heterogeneous effects - Size class differences by number of employees



↓

Download ZEW Discussion Papers from our ftp server:

https://www.zew.de/en/publications/zew-discussion-papers

or see:

https://www.ssrn.com/link/ZEW-Ctr-Euro-Econ-Research.html https://ideas.repec.org/s/zbw/zewdip.html

IMPRINT

ZEW – Leibniz-Zentrum für Europäische Wirtschaftsforschung GmbH Mannheim

ZEW – Leibniz Centre for European Economic Research

L 7,1 · 68161 Mannheim · Germany Phone +49 621 1235-01 info@zew.de · zew.de

Discussion Papers are intended to make results of ZEW research promptly available to other economists in order to encourage discussion and suggestions for revisions. The authors are solely responsible for the contents which do not necessarily represent the opinion of the ZEW.